



Title STEREO- AND REGIOSPECIFIC CATIONIC POLYMERIZATIONS VIA DESIGNED LEWIS ACIDS AND COUNTERANIONS(Dissertation_全文)				
Author(s)	Ouchi, Makoto			
Citation	Kyoto University (京都大学)			
Issue Date	2001-05-23			
URL	http://dx.doi.org/10.14989/doctor.k9097			
Right				
Туре	Thesis or Dissertation			
Textversion	author			



STEREO- AND REGIOSPECIFIC CATIONIC POLYMERIZATIONS VIA DESIGNED LEWIS ACIDS AND COUNTERANIONS

MAKOTO OUCHI

2001

STEREO- AND REGIOSPECIFIC CATIONIC POLYMERIZATIONS VIA DESIGNED LEWIS ACIDS AND COUNTERANIONS

MAKOTO OUCHI

2001

CONTENTS

GENERAL	INTRODUCTION	1
PART I	DESIGN OF INITIATING SYSTEMS FOR STEREOSPECIFIC CATIONIC POLYMERIZAT	ΓΙΟΝ
Chapter 1	Design of Counteranions with	
	Bulky Titanium Lewis Acids as Catalysts	17
Chapter 2	Effects of Monomer Structure	
	on Stereospecificity	35
Chapter 3	Design of Counteranions with	
	Bulky Phosphoric Acids as Initiators	51
PART II	PRECISION CATIONIC POLYMERIZATION	
	OF CYCLOPENTADIENE: CONTROL OF	
	MOLECULAR WEIGHT AND REGIOSPECIFICITY	
Chapter 4	Design of Initiating Systems for	
	Control of Molecular Weight of Poly(CPD)	69
Chapter 5	Design of Initiating Systems for Control of	
	Regiospecificity in CPD Polymerization	87
Chapter 6	Synthesis of End-Functionalized	
	CPD Polymers and Copolymers of CPD	107

LIST OF PUBLICATIONS

ACKNOWLEDGMENTS

127

GENERAL INTRODUCTION

Background

Specificity in **Polymerization.** Chain-growth addition or polymerizations of vinyl monomers (CH₂=CHR) give, in most cases, linear polymers where hundreds and sometimes thousands of repeat units (-CH₂-CHR-) are connected with each other in a regular fashion $-[CH_2-CHR]_n$ (Figure 1). Numerous addition polymerizations are now known, readily giving high polymers in quantitative yield simply by mixing a monomer with an initiator, and this very fact might give an impression that they are easy chemical reactions which do not require serious efforts to control. On the contrary, if one carefully inspects the architectures of any seemingly trivial addition polymers, $-[CH_2-CHR]_n$, one will soon realize that their formation is in fact a result of a chemical process of extremely high specificity and precision.

The linear polymers structure usually consists of a linear chain (or a backbone) of repeat units and terminal groups placed at both ends of the main chain.





-1-

For such a regular backbone to form, the following specificities, among others, should be required:

(a) *Chemospecificity:* The specificity for propagation over any chainbreaking side-reactions, such as chain transfer and termination, which in turn regulates the molecular weight and the molecular weight distribution of the products, along with the structure of the terminal groups. By definition, propagation specificity should be high for high polymers to form, but chainbreaking reactions do occur in most of conventional addition polymerizations. A most important chemospecific chain-growth is generally called as "living polymerization", where the propagation step is free from chain transfer, termination, and any other chain-breaking reactions.¹

(b) Stereospecificity: The specificity for the configuration of repeat units along a backbone, resulting from the stereochemistry of propagation reactions; i.e., the specificity for isotactic, syndiotactic, and heterotactic enchainments in the triad expression (Scheme 1).² A typical example is an iso-

Isotactic



Scheme 1. Common Polymer Tacticities for Mono-Substituted Polymer.

-2-

specific polymerizations.³

(c) *Regiospecificity:* The specificity for the way by which repeat units are connected to each other, resulting from which position of a monomer the growing end attacks: e.g., head-to-head versus head-to-tail structures for a monosubstituted alkene monomer (CH₂=CHR);⁴ 1,2- versus 1,4- structures for a linear^{3,5} or cyclic diene⁶ monomer (CH₂=CH-CH=CHR) (Scheme 2).



Butadiene

1,2-addition

1,4-addition

Scheme 2. Polymerization of Butadiene

Specificity Control in Cationic Polymerization. Among chaingrowth/addition polymerizations of alkenes and related monomers, cationic polymerization seems interesting to the author, because the acid-catalyzed process still involves uncharted frontiers in terms of the precision control of the specificities discussed above.

In 1984 the first examples of living cationic polymerizations of vinyl monomers were reported (cf. the next section),⁷ and since then a series of similar living cationic processes have been developed for almost all of the monomers for carbocationic growth to provide powerful methodologies to synthesize polymers of controlled molecular weights, narrow molecular weight distributions, and well-defined terminal groups.⁸ That implies that high chemospecificity can be attained in cationic polymerization.

In contrast, precision control of stereospecificity and regiospecificity has been particularly difficult in cationic polymerization, except that in almost all cases

-3--

head-to-tail structures form. The current status of these two areas will be reviewed and discussed in each relevant chapter of this thesis, but a brief general overview will follow herein.

Stereospecificity in Cationic Polymerization. In the early phase of research in cationic polymerization, rather accidentally, Schildknecht obtained highly crystalline poly(isobutyl vinyl ether)⁹ that, if property analyzed, would have been the first example of stereoregular synthetic polymers. The polymer was synthesized by a heterogeneous polymerization with boron trifluoride etherate at low temperature, and detailed structure analysis by the original authors and also Natta revealed, shortly after the discovery of isotactic polypropylene,¹⁰ that the crystalline poly(IBVE) is in fact highly isotactic.¹¹ Since then extensive research was carried out in homogeneous¹² and heterogeneous¹³ cationic polymerizations, in part for attaining high stereospecificity and also in relation to the propagation mechanism.¹⁴

These studies revealed, as in other ionic polymerization,¹⁵ that (a) lowering polymerization temperature facilitates stereospecificity;¹² and that (b) isotactic propagation is preferred in nonpolar solvent or other systems where ion-pair intermediates are supposed to prevail.¹⁴ Despite such general understanding, none of the known cationic polymerization systems can achieve such a high and often exclusive stereospecificity for any monomers as in the most sophisticated olefin¹⁶ and anionic polymerizations.¹⁷

Regiospecificity in Cationic Polymerization. Because, almost without exception, cationic polymerization gives regular head-to-tail polymers from mono-substituted monomers, regiospecificity control is to be considered primarily in diene polymerization, namely, the specificity for either the 1,2- or the 1,4-main-chain structures originating from a conjugated allylic carbocation as the propagating species (cf. Scheme 2).¹⁸ As in the stereospecificity control, the general trend in cationic polymerization is similar to those in other ionic counterparts, and the use of nonpolar media at low temperature leads to a moderate

-4--

specificity towards the 1,4-structure for 1-phenylbutadiene¹⁹ and other diene monomers.¹⁸

Another area of interest in terms of regiospecificity control is the cationic polymerization of cyclopentadiene (CPD), a cationically polymerizable cyclic diene that gives polymers of potential applications consisting of thermally stable, completely hydrocarbon cyclic repeat units (Scheme 3). Research in cationic CPD polymerization dates back to 1926 when Staudinger and Brusin reported the formation of 1,2- and 1,4-structures.²⁰ Subsequently some Lewis acids have been reported to polymerize the cyclic monomer into ill-defined polymers.²¹



Scheme 3. Cationic Polymerization of CPD.

Overall, however, for neither linear nor cyclic diene very high 1,2-/1,4regiospecificity has been achieved in cationic polymerization.

Precision Control of Cationic Polymerization via Counteranions. One of the most notable and attractive features in cationic polymerization is the existence of a counteranion accompanying the growing carbocation (Scheme 4). Because the counteranion, by definition, accompanies closely with the growing cationic site, it can affect the course of the propagation reaction, in terms of reactivity (rate), chemospecificity, stereospecificity, and regiospecificity, among other aspects. This, in turn, means that the specificity of propagation reactions can be controlled by designing the counteranions, for example, electronically and sterically.

As illustrated in Scheme 4, a well-defined counteranion can be generated by a combination of a protonic acid (HB; initiator) and a Lewis acid $(MX_n;$

-5--



Scheme 4. Living Cationic Polymerization.

catalyst). The initiator first adds across the double bond of the monomer to form an adduct carrying a covalently attached anionic part (B) of the acid. This linkage is ionically dissociated by the Lewis acid or the catalyst, and thereby a counteranion ($-B-MX_n$) is formed. Thus, the nature of the growing species ($--CH_2-CHR^+$ $-B-MX_n$) will be governed by the substituent R from the monomer, the anionic part B from the initiator, and the Lewis acid (MX_n) as the catalyst; note that the last two components give the counteranion $-B-MX_n$.

In fact, the fine chemospecificity control, or living cationic polymerization, has been achieved by such two component initiating systems (HB/MX_n) (see above), and, in these examples, the nucleophilicity of the counteranions is considered the most critical factor for the chemospecificity.⁸ Unfortunately, however, none of such general principles have been found for precision control of stereo- and regiospecificity in cationic polymerization.

Objectives: Stereo- and Regiospecific Cationic Polymerizations

From these backgrounds, the author focused his research on the "stereospecific" and "regiospecific" cationic polymerizations. As described above, there have few reports to prepare high-stereoregular polymers or regiospecific

-6-

polymers in cationic polymerization. He believed that some particular designed initiating system should be required for the control. In view of these, twocomponent initiating systems [a protonic acid (HB) and a Lewis acid (MX_n)] will be suitable to achieve such selectivities via the design of counteranion from steric and electronic viewpoints (Scheme 5).



Scheme 5. Environment around Growing Carbocationic Species in Cationic Polymerization of VE and CPD.

Therefore, the first object of this study is concerned with stereospecific cationic polymerization of vinyl ethers by a systematic search of the three factors: Lewis acids, monomer substituents, and initiators. Especially, modification of Lewis acid is most interesting, because it can be sterically and electronically designed to give unique environment in the growing end. Thus, the author employed various bulky titanium compounds in place of conventional Lewis acids or metal halides for the cationic polymerization to examine the possibility of stereoregular cationic polymerization. For this purpose, he selected alkyl vinyl ethers (VEs) as monomers, which show high-reactivity in cationic polymerization and then even the weaker designed Lewis acids with some substituents can be used as well. The effects of alkyl group of monomer were also examined due to possibilities of the interaction between oxygen of monomer and central metal of

Lewis acid. Along with these efforts, he also surveyed the bulky protonic acids as initiators.

Secondly, in connection with regioselectivity, the author noted the cationic polymerization of cyclopentadiene (CPD) because the growing species is most likely a resonating allylic carbocation and thus leads to the regioselectivity, namely 1,2- and 1,4-adducut. To this end, he examined effects of the various factors on regioselectivity. Prior to the examination of the regioselectivity, he decided to develop the living or controlled cationic polymerization of CPD which has never been achieved.

Outline of This Study

The present thesis consists of two parts: **Part 1** (Chapters 1–3) deals with the stereospecific cationic polymerization with designed counteranions derived from a protonic acid (initiator) and a Lewis acid (activator/catalyst). Effects of monomer substituents are also discussed. **Part II** (Chapters 4–6) discusses the controlled and regiospecific cationic polymerization of cyclopentadiene (CPD) terms of molecular weight and microstructure. The initiating systems were then applied for the syntheses of various CPD-based precision polymers.

Chapter 1 presents the iso-specific cationic polymerization of isobutyl vinyl ether (IBVE) with designed bulky titanium compounds. A series of titanium dichloride $[TiCl_2(OAr)_2]$ were also employed where the metal center carries two substituted phenoxy groups, and a pronounced dependence of isotacticity was observed on the bulkiness, positions, and structure of ring-substituents on OAr (Scheme 6). Overall, isopropyl groups at 2,6-positions on the phenyl group Ar led to the highest isotacticity, whereas bulkier or planer-like groups such as *tert*-butyl or phenyl were less effective, suggesting that not only bulkiness but three-dimensional shapes of Lewis acids are critical in stereoregulation in cationic

-8-



Scheme 6. Cationic Polymerization of IBVE with Bulky Lewis Acids.

polymerization.

Chapter 2 focuses on the substituents on the vinyl ether monomers with the titanium compounds described in Chapter 1. The tacticities depend on the substituents, where isobutyl and isopropyl vinyl ethers gave highly isotactic polymer (mm = 83%), whereas *tert*-butyl vinyl ether (TBVE) resulted in a lower isotactic content (mm = 49%) similar to that with a precursor, TiCl₄. The statistical analysis revealed that the high isospecificty is achieved not by the chain end but by the catalyst or the counteranion derived therefrom.

In **Chapter 3** the author focused on the tacticity control by designed bulky counteranions derived from initiators (HB or B^-) for IBVE cationic polymerization. Therefore, phosphoric acid derivatives [(RO)₂POOH] coupled with SnCl₄ gave



Scheme 7. Cationic Polymerization of IBVE with Bulky Phosphoric Acids.

-9-

highly isotactic poly(IBVE)s, where those with bulky substituent (R), $[C_4H_9CH(C_2H_5)CH_2O]_2POOH$ and $(n-C_{10}H_{21})_2POOH$, led to the highest isotacticity (m = 86%) (Scheme 7). In contrast, isospecificity was lower with IBVE-HCl and CF₃COOH under the same conditions.

Chapter 4 presents the first example of a precision cationic polymerization of cyclopentadiene (CPD) that forms controlled poly(CPD) with narrow molecular weight distributions (MWDs). With the use of HCl-adduct of CPD or a vinyl ether (initiator), SnCl₄ (Lewis acid as catalyst), and *n*-Bu₄NCl (additive), the number-average molecular weights of the polymers increased in direct proportion to monomer conversion, as well as to the monomer-to-initiator mole ratio, and the MWDs were narrow throughout the reactions ($M_w/M_n =$ 1.2–1.3), clearly narrower than those in the additive-free systems ($M_w/M_n > 2$) (Scheme 8). The prepared poly(CPD) carried the initiator fragment at the α -end (nearly one per chain).



Additive: n-Bu₄NCI, Et₂O, CH₃CO₂Et₂

Scheme 8. Controlled Cationic Polymerization of CPD.

Chapter 5 focuses on the regiospecific polymerization of CPD with the binary initiating systems consisting of a protonic acid and a Lewis acid. The reaction parameters, e.g., solvents, additives, and temperatures, did not dramatically affect regioselectivity, giving 55–60% 1,4–contents almost invariably. On the other hand, a Lewis acid has dramatic effects on the selectivity, where weak

Lewis acids such as ZnX_2 (X = Cl, Br, I) gave the highest [1,4]-contents (64, 70, 76%, respectively) (Scheme 9). In particular, the CPD-HCl/ZnBr₂ system induced controlled polymerization to give poly(CPD)s with narrow molecular weight distributions (MWDs) ($M_w/M_n = 1.3-1.5$) and relatively high regiospecificity ([1,4] = 70%).



MX_n: SnX₄ (X = Cl, Br), TiX₄ (X = Cl, Br), TiCl₂(O*i*-Pr)₂, AlBr₃, BCl₃, ZnX₂ (X = Cl, Br, I)

Scheme 9. Regioselectivity of CPD Polymerization with Lewis Acids

Chapter 6 is directed toward the syntheses of a series of CPD-based polymers and copolymers by the controlled systems showed in chapter 4 (Scheme 10). With HCl-adducts of vinyl ethers carrying pendant functional substituents as initiators, end-functionalized poly(CPD) was synthesized in conjunction with SnCl₄. The system led to controlled cationic polymerizations of CPD to give controlled α -end-functionalized poly(CPD)s with almost quantitative attachment of the functional groups ($F_n \sim 1$). Additionally, diblock copolymers of 2-chloroethyl vinyl ether (CEVE) and 2-acetoxyethyl vinyl ether with CPD were also synthesized by sequential polymerization of CPD and these vinyl ethers. ABA-Type triblock copolymer of CPD (A) and CEVE (B) was also prepared with the use of a bifunctional initiator.



Scheme 10. Syntheses of CPD-Based Polymers and Copolymers

In conclusion, this thesis presents the control of stereo- and regioselectivity in cationic polymerization by designed counteranions, intensively Lewis acids. Also the controlled polymerization of CPD and its application are discussed. The author hopes that this thesis contributes to further developments in not only cationic polymerization but also novel controlled organic reactions via cationic intermediates.

REFERENCES

- For Recent Reviews, see: (a) Szwarc, M. Carbanions, Living Polymers, and Electron-Transfer Processes, Interscience, New York, 1968. (b) Webster, O. W. Science 1991, 251, 887. (c) Aida, T. Prog. Polym. Sci. 1994, 19, 469.
- (2) IUPAC Pure Appl. Chem. 1996, 68, 2287.
- (3) Coates, G. W. Chem. Rev. 2000, 100, 1223.
- (4) (a) Doi, Y. Macromolecules 1979, 12, 248. (b) Pino, P.; Mulhaupt, R. Angew. Chem., Int. Ed. Engl. 1980, 19, 857. (c) Park, J. R.; Shino, T.; Soga, K. Macromolecules 1992, 25, 521.

- (5) Watanabe, H.; Masuda, N. In Catalysis in Precision Polymerization, Kobayashi, S.; Wiley, Chichester, England, Chapter 2, 1997.
- (6) (a) Natori, I. Macromolecules 1997, 30, 3696. (b) Natori, I.; Imaizumi, K.;
 Yamagishi, H.; Kazunori, M. J. Polym. Sci. Part B: Polym. Phys. 1998, 36, 1657.
- Miyamoto, M.; Sawamoto, M.; Higashimura, T. Macromolecules 1984, 17, 265.
- (8) For recent reviews on living cationic polymerization, see: (a) Sawamoto, M. Prog. Polym. Sci. 1991, 16, 111. (b) Kennedy, J. P.; Iván, B. Designed Polymers by Carbocationic Macromolecular Engineering: Theory and Practice; Hanser: Munich, 1992. (c) Matyjaszewski, K., Ed.; Cationic Polymerizations, Marcel Dekker: NY, 1996.
- (9) Schildknecht, C.E.; Zoss, A. O.; McKinley, C. Ind. Eng. Chem. 1947, 39, 180.
- (10) Natta, G.; Pino, P.; Corradini, P.; Danusso, F.; Mantica, E.; Mazzanti, G.; Moraglio G. J. Am. Chem. Soc. 1955, 77, 1708.
- (11) Natta, G. J. Polym. Sci. 1960, 48, 219.
- (12) (a) Higashimura, T.; Yonezawa, T.; Okamura, S.; Fukui, K. J. Polym. Sci. 1959, 39, 487. (b) Yuki, H.; Hatada, K.; Ota, K.; Kinoshita, I.; Murahashi, S.; Ono, K.; Ito, Y. J. Polym. Sci., A-1 1969, 7, 1517. (c) Sikkema, D. J.; Angad-Gaur, H. Makromol. Chem. 1980, 181, 2259. (d) Angad-Gaur, H.; Sikkema, D. J. Makromol. Chem. 1980, 181, 2385. (e) Ohgi, H.; Sato, T. Macromolecules 1993, 26, 559. (f) Ohgi, H.; Sato, T. Macromolecules 1999, 32, 2403.
- (13) Okamura, S.; Higashimura, T.; Watanabe, T. Makromol. Chem. 1961, 50, 137.
- (14) Kunitake, T.; Aso, C. J. Polym. Sci., A-1 1970, 8, 665.
- (15) Hatada, K.; Kitayama, T.; Ute, K. Prog. Polym. Sci. 1988, 13, 189.

-13-

- (16) For recent reviews, see: (a) Ziegler Catalysts; Fink G., Mülhaupt R., Brinzinger H. H., Eds.; Springer-Verlag: Berlin, 1995. (b) Brinzinger, H. H.; Fischer, D.; Mülhaupt, R.; Rieger, B.; Waymouth, R. M. Angew. Chem. Int. Ed. Engl. 1995, 34, 1143.
- (17) For a recent review, see: Hatada, K.; Kitayama, T. In Macromolecular Design of Polymeric Materials; Hatada, K., Kitayama, T., Vogl, O., Eds.; Marcel Dekker: New York, 1997; p. 139.
- (18) For a review on cationic polymerization of conjugated dienes; see: Kennedy,
 J. P. Cationic Polymerization of Olefins: A Critical Inventory, Wiley-Interscuence, New York, 1975, p. 154.
- (19) Radzitski, P.; Wilde, M. C.; Smets, G. J. Polym. Sci. 1954, 13, 477.
- (20) Staudinger, H.; Bruson, H. A. Ann. 1926, 447, 110.
- (21) (a) Sigwalt, P.; Vairon, J. P. Bull. Soc., Chim. France 1964, 482. (b) Momiyama, Z.; Imanishi, Y.; Higashimura, T. Chem. High Polymers, 1966, 23, 56. (c) Aso, C.; Kunitake, T.; Ishimoto, Y. J. Polym. Sci. Part A-1 1968, 6, 1163. (d) Aso, C.; Kunitake, T.; Ishimoto, Y. J. Polym. Sci. Part A-1 1968, 6, 1175.

PART I

DESIGN OF INITIATING SYSTEMS FOR STEREOSPECIFIC CATIONIC POLYMERIZATION

Chapter 1

Design of Counteranions with Bulky Titanium Lewis Acids as Catalysts

ABSTRACT

Highly isotactic-specific cationic polymerization has been achieved for isobutyl vinyl ether (IBVE) with bis[(2,6-diisopropyl)phenoxy]titanium dichloride $[TiCl_2{OC_6H_4(i-Pr)_2(2,6-)}_2(2)]$ as a Lewis acid catalyst (activator) to be coupled with the IBVE-HCl adduct (initiator). The polymerization, homogeneous and quantitative, was carried out in *n*-hexane at -78 °C in the presence of a bulky pyridine (2,6-di-tert-butyl-4-methylpyridine) to give isotactic-rich polymers with a meso dyad (m) = 90–92% ($M_n \sim 20000$). Under similar conditions unsubstituted counterparts (TiCl₄, SnCl₄, etc.) gave nearly atactic products (m = 68%). With catalyst 2, the isotactic content m increased at lower temperature, in less polar solvents, and with the added pyridine (m = 64-92%). To design Lewis acids for stereoregulation, a series of titanium dichloride [TiCl2(OAr)2] were also employed where the metal center carries two substituted phenoxy groups, and a pronounced dependence of isotacticity was observed on the bulkiness, positions, and structure of ring-substituents on OAr. Overall, isopropyl groups at 2,6-positions led to the highest isotacticity, whereas bulkier or planer-like groups such as tert-butyl or phenyl were less effective, suggesting that not only bulkiness but threedimensional shapes of Lewis acids are critical in stereoregulation in cationic polymerization.

INTRODUCTION

Stereoregularity is a factor that seriously affects physical properties of polymers. Thus stereospecific polymerizations have extensively been explored and are indeed possible for various monomers. Examples include coordination polymerization of olefins with Zieglar-Natta or metallocene catalysts¹ and anionic polymerization of methacrylates with *t*-BuMgBr and *t*-BuLi/alkyl aluminum.²

In cationic polymerization, however stereoregularity control has been difficult, particularly in homogeneous systems, although nearly complete control of molecular weight has been achieved by various living systems.³ Effects of catalysts or counteranions, substituents of monomers, solvent polarity, and temperature on tacticity have in fact been investigated rather extensively in cationic polymerization of vinyl ethers (VE), and mechanisms have been proposed for the formation of isotactic and syndiotactic enchainments,^{4,5} but very high stereospecificity is still beyond reach thus far.

For example, it has been reported that BF₃OEt₂,⁶ AlEtCl₂,⁷ and AlEt₂Cl⁷ give more or less isotactic poly(VE) in less polar solvents at low temperature, but the stereoregularity is not very high ($mm \sim 70\%$). Vinyl ethers with bulky substituents like benzyl⁸ and *tert*-butyl⁹ were polymerized with BF₃OEt₂ to afford polymers of relatively high isotacticity [mm = 86% (benzyl) and 79% (*tert*-butyl)]. Allyl vinyl ether has also been polymerized by SbCl₅ into isotactic polymer with m = 80%.¹⁰ In addition, heterogeneous systems with aluminum sulfate and sulfuric acid give crystalline stereoregular poly(VE), where the control of stereoregularity has been made by the orientation and absorption of monomers on the surface of the catalysts.¹¹ However, the tacticity of the polymers was unknown because of the lack of NMR analysis at that time.

More recently, Sawamoto, Kamigaito, and their co-workers have developed a series of living cationic polymerizations of vinyl ethers, styrene derivatives, and related monomers (eq 1), which are almost invariably initiated by binary systems

-18-



B: I, Br, CI, OCOCF₃, etc. MX_n: ZnCl₂, SnBr₄, TiCl₂(O*i*Pr)₂, etc.

consisting of a protonic acid (HB; initiator) and a Lewis acid (MX_n) . Lewis acid acts as a catalyst/activator that interacts with the terminal covalent bonds (\sim C–B) derived from initiator HB and thereby induces its reversible dissociation into a growing carbocation. Via this dissociation, in turn, the Lewis acid combines with the anionic part(B) of the terminal to form the counteranion $B-MX_n^{-}$, which associates with the growing carbocation, by one way or another, and thus affects the nature of the active species. In the course of these studies, they have in fact noted that the judicious choice of the Lewis acids is crucial in achieving living cationic polymerization, or precise control of the propagation process. In particular, a series of titanium chlorides and alkoxides, $Ti(OR)_n Cl_{4-n}$ (R = alkyl or aryl) are quite useful in that, by changing the number (n) and the structure of the alkoxyl or aryloxyl substituents (OR), one can systematically control (and thus design) their Lewis acidity and in turn the nucleophilicity of the counteranions.¹² For vinyl ethers, the dichlorides such as TiCl₂(Oi-Pr)₂ and TiCl₂(OPh)₂ proved most efficient and of suitable Lewis acidity to effect living cationic polymerization.

From these findings the author then notes, relative to the stereoregulation of cationic polymerization, that such feature of the HB/MX_n binary initiating systems would be advantageous, because one could control the stereochemical environment of the growing carbocation through designing the Lewis acid components that

Chapter 1

determine the bulkiness, topology, and electronic nature of the associating counteranion. To his knowledge, such a Lewis acid-based approach to the counteranion design is unpreceded, neither in efforts for stereoregulation of cationic polymerization nor in fundamental studies of its stereochemistry. For example, in their now classical work, Higashimura⁴ and Kunitake⁵ were indeed aware of the importance of counteranions in determining, at least in part, the steric course of vinyl ether polymerizations, but their primary attention was apparently directed to the ionic radii and average charge density of counteranions. Therefore, they mostly employed simple spherical counteranions derived from a series of metal halides where the central metals alone were systematically varied. However, these studies led to important knowledge that isotactic propagation is preferred in ion-pair growing ends, i.e., in nonpolar solvent and with suitably bulky counteranions.

In this chapter, first of this thesis, the author therefore examined the possible stereoregulation of cationic polymerization by employing a series of titanium dichlorides $[TiCl_2(OAr)_2]$ for which the bulkiness, topology (spatial shape), and electronic nature of counteranions can be manipulated by designing the structure of the aromatic substituents OAr (eq 2). Herein he employed a series of bulky titanium compounds, 2,6-disubstituted TiCl₂(OAr)₂, for the cationic polymerization of isobutyl VE (IBVE) in conjunction with adduct 1 (eq 2). His preliminary results show that the tacticity of poly(IBVE) depends on the



-20-

substituents of the titanium compounds¹³. The author reports here that bis(2,6-diisopropylphenoxy) titanium dichloride gives highly isotactic poly(IBVE) (m = 90-92%, mm = 83%) in *n*-hexane at -78 °C in the presence of a bulky pyridine additive (2,6-di-*tert*-butyl-4-methylpyridine: DTBMP).

RESULTS AND DISCUSSIONS

1. Cationic Polymerization of IBVE with Substituted TiCl₂(OAr)₂

IBVE was polymerized with various 2,6-disubstituted phenoxy titanium(IV) dichlorides $[TiCl_2(OAr)_2]$ in conjunction with the HCl-adduct of IBVE (1) in toluene at -78 °C. The ring substituents have been selected in terms of their bulkiness and shape, along with their electronic nature. Figure 1 shows



Figure 1. Polymerization of IBVE with $1/\text{TiCl}_2(\text{OAr})_2$ in toluene at -78 °C: $[M]_0 = 0.38$ M; $[1]_0 = 5.0$ mM; $[\text{TiCl}_2(\text{OAr})_2]_0 = 10$ mM. 2,6-substituent (R) on OAr: R = Cl (O); *i*Pr (\oplus); Ph (\Box).

-21-

conversion-time curves of the polymerizations and size-exclusion chromatograms (SEC) of the polymers obtained with various titanium compounds. The polymerization rate depends on the substituents; with the 2,6-dichlorophenoxy titanium compound, which has electron-withdrawing chloro atoms on the phenoxy group, the polymerization occurred very fast and completed in 1 min. On the other hand, the polymerization was very slow with the corresponding 2,6-diphenylphenoxide, which has two bulky substituents on the phenoxy group. These results show that the nature of 2,6-substituents on the phenoxy groups of the titanium compounds has great effects upon the polymerizations.

The number-average molecular weights (M_n) were larger than the calculated values, and the molecular weight distributions (MWDs) were broad. These titanium compounds with substituted phenoxide groups were not effective in living polymerization. Such substituents on the phenoxy group near the titanium center may prevent effective interaction between the terminal chloride and the metal due to the steric hindrance, which results in slow interconversion between the dormant and the active species and broad MWDs.

2. Steric Structure of Poly(IBVE) with TiCl₂(OAr)₂

As shown in the representative ¹³C NMR spectrum of a sample obtained with TiCl₂{OC₆H₃*i*-Pr₂(2,6-)}₂ (**2**), the dyad distributions of poly(IBVE) can be determined from the peak intensity ratios of the main-chain methylene carbons $[-CH_2-CH(OiBu)-]$ around 39–42 ppm where the meso (*m*) and racemo (*r*) dyads are well resolved (Figure 2).¹⁴ The absorptions of the methylene carbons $[-OCH_2-CH(CH_3)_2]$ around 76–77 ppm can be used for the determination of triad distributions, but the signals of CDCl₃ at 77 ppm prevent the precise integrations of these peaks. Spectra of some samples were thus recorded in C₆D₆ especially for the determination of triad distributions (see below).

Figure 3 compares the partial ¹³C NMR spectra for the main-chain methylene signals for a series of poly(IBVE) samples obtained in toluene with

-22-



Figure 2. ¹³C NMR spectrum (67.9 MHz) of poly(IBVE) in CDCl₃ at 55 °C. Polymerization conditions: $[M]_0 = 0.38$ M; $[1]_0 = 5.0$ mM; $[TiCl_2{OC_6H_3i-Pr_2(2,6-)}_2]_0 = 10$ mM, in toluene at -78 °C.

titanium tetrachloride and four TiCl₂(OAr)₂ derivatives with 2,6-disubstituted phenoxy groups [OAr = O-C₆H₃Y₂(2,6-); Y = C₆H₅, *tert*-Bu, Cl, *i*-Pr]. Except for the difference in Lewis acids, all of the samples were prepared under the same conditions (see Figure 1, caption). The dyad meso/racemo (*m*/*r*) peak ratios and the peak width clearly changed with the Lewis acids. Namely, in all cases the meso signal is predominant over the racemo counterpart, but meso dyad was rather low (72 %) of the unsubstituted form (TiCl₄) and even for the derivatives with very bulky ring-substituents (Y = C₆H₅, *tert*-Bu) (Figure 3, A–C). Thus, the introduction of bulky groups on the 2,6-positions of the OAr seems not generally effective, relative to TiCl₄, in controlling stereochemistry of the vinyl ether propagation.

In contrast, the isopropyl form (Y = i-Pr; Figure 3, E), whose 2,6substituents are less bulkier than phenyl or *tert*-butyl, gave clearly larger and

-23-



Figure 3. ¹³C NMR spectra (67.9 or 125.8 MHz, CDCl₃, 55 °C) around 39–42 ppm of poly(IBVE) obtained with $1/MX_n$ at -78 °C: (A); $[M]_0 = 0.38$ M, $[1]_0 = 5.0$ mM, $[TiCl_4]_0 = 2.0$ mM, in toluene, (B)–(E); $[M]_0 = 0.38$ M, $[1]_0 = 5.0$ mM, $[MX_n]_0 = 10$ mM, in toluene.

sharper meso signals to give m = 83%, the highest among the seven TiCl₂(OAr)₂type Lewis acids employed herein under these conditions (see below also). When compared with those for TiCl₄ and the phenyl or *tert*-butyl counterparts, the spectral changes are much clearer than that in the rather insensitive numerical meso values: the three meso signals for the isopropyl form are sharper; the *mmm* peak is particularly predominant; and the racemo peak is very small.

These trends demonstrate that the 2,6-disubstituted phenoxy rings are effective in facilitating the meso propagation but that the bulkiness of the substituent Y [in $-OC_6H_3Y_2(2,6-)$], or the OAr groups in general, is not a primary



Figure 4. Meso dyad contents of poly(IBVE) obtained with $1/MX_n$ in toluene at -78 °C and computer-generated molecular models of titanium compounds. Polymerization conditions: $[M]_0 = 0.38$ M; $[1]_0 = 5.0$ mM; $[TiCl_2(OAr)_2]_0 = 10$ mM (2.0 mM only for TiCl₄).

factor. This point can be better illustrated in Figure 4, which shows the computergenerated space-filling molecular models of the seven $TiCl_2(OAr)_2$ derivatives and their parent form TiCl₄, along with the meso contents of the polymer samples obtained from them (under the same conditions as for those in Figure 3). The models are generated by a molecular mechanics program (MM2) where the conformation is optimized in bond length and bond angle so as to minimize strain energy. In the figure they are shown edge-on in terms of the dihedral plane for the Cl–Ti–Cl bonds; the Ti center is seen behind the two chlorines. In all cases the TiCl₂ part is fairly well exposed for the interaction with the terminal chlorine in generating the carbocationic growing species (cf. Scheme 1).

Obviously, the titanium center becomes more sterically hindered with bulkier 2,6-substituents (Y) are attached on the phenoxy groups. Rather unexpectedly, however, the highly bulky forms with Y = phenyl or *tert*-butyl resulted in rather low meso contents (m = 72%) without appreciable changes from the results for the least bulky parent TiCl₄. Along with another bulky derivative with a bridged bisphenoxy group (m = 70%), these crowded Lewis acids appear to form counteranions too bulky for effective stereoregulation. Note that they in fact induced very slow polymerizations only (cf. Scheme 1), probably too bulky to effectively interact with the terminal chlorine, either.

The diisopropyl form (2), which gave the highest meso content among the seven acids, appears less bulky (Figure 4, top left). However, inspection of its molecular model indicates that the isopropyl groups are twisted relative to the aromatic ring, and because of this the space around the titanium center is even more sterically hindered than in the *tert*-butyl or phenyl derivatives. Apparently similar steric environment can be seen for the 2-*tert*-butyl-6-methyl form, which also gave relatively high meso value (78%).

These results suggest that stereoregulation is affected by not only overall bulkiness of the Lewis acid or its substituents but other factors, such as the spatial shape and symmetry around the Lewis acid's metal center from which the

-26-

counteranion is formed by attaching a chlorine of the dormant terminal. The latter spatial factor is therefore expected to affect the bulkiness and spatial shape of the resulting counteranions, which interact with the cationic growing site and is directly responsible for stereoregulation. Further studies are needed in this line.

3. Highly Isotactic Poly(IBVE): Effects of Reaction Conditions

The steric course of propagation will be affected by interaction of the counteranion with the growing carbocation, and the interaction, in turn, will be a function of not only the counteranion structure but also reaction conditions such as solvent polarity and temperature. The author thus examined the effects of these conditions on the IBVE polymerization with the Lewis acid 2 [TiCl₂{OC₆H₃*i*-Pr₂(2,6-)}₂] which proved most isotactic-selective among the seven Lewis acids so

_									
	entry	temp.	solvent	additive	conv.	M _n	$M_{\rm w}/M_{\rm n}$	meso	
		°C			%			%	
	1	-78	CH ₂ Cl ₂	none	98	10,100	2.56	82	
	2	-78	toluene	none	99	38,300	3.54	83	
	3	-78	<i>n</i> -hexane	none	93	23,800	3.43	86	
	4	-78	toluene	DTBMP	95	36,400	2.32	89	
	5	-78	n-hexane	DTBMP	95	47,400	1.93	90	
	6 ^b	-78	n-hexane	DTBMP	80	19,400	2.56	92	
	7	-15	CH_2Cl_2	DTBMP	98	5,800	1.28	64	
	8	-15	CH_2Cl_2	DTBMP	98	12,100	1.12	71	

Table 1.	Effects	of Solvent	Polarity,	Additive,	and,	Temperature
	on Ca	tionic Poly	merizatio	on of IBVE	E with	2^{a}

^{*a*} $[M]_0 = 0.38 \text{ M}; [1]_0 = 5.0 \text{ mM}; [2]_0 = 10 \text{ mM}; [DTBMP]_0 = 0 \text{ or } 2.0 \text{ mM}.$ ^{*b*} The polymerization was carried out via addition of IBVE into the mixture of 1, 2, and DTBMP in *n*-hexane at -78 °C.

-27-

Chapter 1



Figure 5.

¹³C NMR spectra (67.9 or 125.8 MHz, CDCl₃, 55 °C) around 39–42 ppm of poly(IBVE) obtained with 1/2 at -78 °C: (A); $[M]_0$ = 0.38 M, $[1]_0$ = 5.0 mM, $[2]_0$ = 10 mM, in CH₂Cl₂, (B); $[M]_0$ = 0.38 M, $[1]_0$ = 5.0 mM, $[2]_0$ = 10 mM, in toluene, (C); $[M]_0$ = 0.38 M, $[1]_0$ = 5.0 mM, $[2]_0$ = 10 mM, in *n*-hexane, (D); $[M]_0$ = 0.38 M, $[1]_0$ = 5.0 mM, $[2]_0$ = 10 mM, in *n*-hexane, (D); $[M]_0$ = 0.38 M, $[1]_0$ = 5.0 mM, $[2]_0$ = 10 mM, in *n*-hexane, (D); $[M]_0$ = far discussed. The results are summarized in Figure 5 (13 C NMR; the main-chain methylene region) and Table 1 (meso dyad data).

As in the usual cationic polymerization, the meso contents increased with decreasing solvent polarity ($CH_2Cl_2 < toluene < n$ -hexane; Figure 5, A–C; Table 1, entries 1–3), and the value for hexane exceeded that for toluene (86% versus 83%). This is due to the suppression of ionic dissociation of the growing carbocation.

Addition of 2,6-di-*tert*-butyl-4-methylpyridine (DTBMP), known as a hindered base or as a proton trap, increased the meso dyad value, especially 90% in n-hexane (Table 1, entry 5 and Figure 5D). The effects of DTBMP is probably ascribed to the suppression of the initiation from the adventitious water that would generate hydroxide-based less nucleophilic counteranions or to possible interaction between DTBMP and the growing species.

All the polymerizations thus far investigated were initiated by the addition of 2 into the solution of monomer and 1 in *n*-hexane at -78 °C. To diminish the polar effects of the vinyl ether monomer, the author also examined the polymerization initiated by slow addition of IBVE into the solution of 1 and 2 in *n*hexane at -78 °C (Table 1, entry 6). This procedure gave higher meso content (92%). The triad distribution (*mm*) was 83%, the highest value for poly(IBVE) to our knowledge.

Effects of temperature were investigated where the polymerizations were carried out at -15 °C in the presence of DTBMP (Table 1, entries 7 and 8). The stereoregularity is decreased by increasing the temperature as in usual ionic polymerizations. However, the molecular weight distributions were narrower, and the molecular weight increased with conversion, which suggests occurrence of living polymerization.

4. DSC Analysis of the Isotact-Rich Poly(IBVE)

The formation of isotact-rich poly(IBVE) with $m \ge 90\%$ promoted him to investigate the thermal property of the polymer by differential scanning

-29-

calorimetry (DSC). The author thus used the isotact-rich polymer (Table 1, entry 6: m = 90%, $M_n = 32100$, $M_w/M_n = 2.12$) and the atactic polymer (m = 68%, $M_n = 33200$, $M_w/M_n = 1.16$) obtained by the living cationic polymerization 1/SnCl₄/nBu₄NCl systems at -15 °C in CH₂Cl₂,¹⁵ where both the M_n were about 3 × 104. Both samples showed an endothermic bias at -14 °C due to glass transition, whereas another endothermic peak, probably arising from melting, was different in the two; the isotact-rich poly(IBVE) showed this peak at 131°C whereas the atactic poly(IBVE) at 47 °C. There is also difference in appearance, where the former was stiff film and the latter was soft semisolid. These results show that soft and adhesive character of poly(VE) can be changed by tacticity of polymers, which may lead to new character of poly(VE) as materials.

In conclusion, control of stereoregularity in cationic polymerization can be achieved by the Lewis acid-type activators based on their substituents. Especially, bis(2,6-diisopropylphenoxy)titanium dichloride (2) gave isotact-rich ($m \ge 90\%$) poly(IBVE) in *n*-hexane at -78 °C in the presence of DTBMP in conjunction with 1.

EXPERIMENTAL

Materials

IBVE (Tokyo Kasei; purity > 99%) was washed with 10% aqueous sodium hydroxide and then with water, dried overnight over potassium hydroxide (pellets), and distilled twice over calcium hydride before use. Toluene, *n*-hexane, and CH₂Cl₂ (solvents) and carbon tetrachloride (an internal standard for gas chromatography) were purified by the usual methods and distilled twice over calcium hydride before use. TiCl₄ (Aldrich, > 99%) was used as received. 2,6diisopropyl phenol (Tokyo Kasei, > 98%), 2,6-di-*tert*-butyl phenol (Wako, > 98%), 2,6-dichloro phenol (Aldrich, > 99%), 2,6-diphenyl phenol (Aldrich, > 98%), 2tert-butyl-6-methyl phenol (Tokyo Kasei, > 95%), and 2,2'-methylene-bis (6-tertbutyl-4-ethyl) phenol (Aldrich, > 90%) were used as received for the preparation of $TiCl_2(OAr)_2$ as below. All the 2,6-disubstituted phenoxy titanium(IV) dichlorides [$TiCl_2(OAr)_2$] except $TiCl_2(OPh)_2$ were prepared as below. $TiCl_2(OPh)_2$ was prepared and purified as already reported.¹²

Preparation of TiCl₂(OAr)₂

All the procedures were carried out under dry argon in baked glassware equipped with three-way stopcocks. All the solvents were used after purification All the 2,6-disubstituted phenoxy titanium(IV) dichlorides and distillation. [TiCl₂(OAr)₂] were prepared by stirring titanium(IV) chloride and corresponding 2 equivalent 2,6-disubstituted phenol under argon flow to remove the produced HCl gas. For example, 2 was prepared as below. Titanium(IV) chloride (2.6 mL, 23.7 mmol) was added dropwise to a solution of 2,6-diisopropyl phenol (8.8 mL, 47.2 mmol) via a dry syringe in CH₂Cl₂ (30 mL) at -30 °C. On addition of TiCl₄, HCl gas was generated, which was trapped by bubbled into 10% NaOH solution. After stirring at -30 °C for 10 min, the mixture was stirred at room temperature for After the solution was evacuated by vacuum pump, the reddish brown liquid 1 h. remained and then dissolved in toluene. The solution was sealed in brown ampules and stored in the dark in a freezer. 2 was also prepared by another method where equimolar TiCl₄ and Ti(OAr)₄ was mixed together.

Synthesis of HCl–IBVE Adduct (1)

The HCl–IBVE adduct 1 was synthesized by bubbling dry HCl gas through a solution of IBVE (9.0 mL IBVE in 60 mL *n*-hexane; 1.0 M) at -78 °C. The gas was generated by dropping concentrated sulfuric acid into powdery sodium chloride, and dried by passing it through a column packed with calcium chloride. After 30 min, the excess HCl in the reaction mixture was removed by bubbling dry

-31-

Chapter 1

nitrogen gas at 0 °C for 30 min, and the clean and quantitative formation of adduct 1 was confirmed by ¹H NMR spectroscopy. The concentration of 1 was determined by titrating the chloride by the Volhard method; the observed value (~ 1 M) further confirmed the quantitative formation of 1 from IBVE.

Polymerization Procedures

Polymerization was carried out under dry nitrogen in baked glass tubes equipped with a three-way stopcock. A typical example for the polymerization procedure in toluene is given below: The reaction was initiated by sequential addition of prechilled solutions of 1 (in *n*-hexane; 0.30 mL) and titanium complex (in toluene, 0.30 mL) via dry syringes into a monomer solution (in toluene; 2.4 mL) containing IBVE (0.15 mL) and CCl₄ (0.15 mL). After predetermined intervals, the polymerization was terminated with prechilled methanol (2.0 mL) containing a small amount of ammonia. Monomer conversion was determined from its residual concentration measured by gas chromatography with CCl₄ as an internal standard. The polymer yield by gravimetry was in good agreement with the gas-chromatographic conversion of the monomer.

The quenched reaction mixture was washed with dilute hydrochloric acid, aqueous sodium hydroxide solution, and then with water to remove initiator residues, evaporated to dryness under reduced pressure, and vacuum dried to give the product polymers.

Measurements

The MWD, M_n , and M_w/M_n ratios of the polymers were measured by sizeexclusion chromatography (SEC) in chloroform on three polystyrene gel columns (Shodex K-805L × 3) that were connected to a Jasco PU–980 precision pump and a Jasco 930-RI refractive index detector. The columns were calibrated against 11 standard polystyrene samples (Pressure Chemicals; $M_n = 580-1547000$; $M_w/M_n \le$ 1.1) as well as the monomer. ¹H and ¹³C NMR spectra of the produced polymers

-32-
were recorded in CDCl₃ or CCl₄/C₆D₆(9/1 v/v) at 55 °C on a JEOL LNM–LA500 or a JEOL JNM–GSX270 spectrometer, operating at 500.2 or 270.2 MHz (¹H) and 125.8 or 67.9 MHz (¹³C), respectively. The DSC measurements were performed on a Thermal analyzer system WS002 (MAC Science, Tokyo) equipped with a TAPS1000S control unit and a DSC3100S module. Polymer solid samples (ca. 8 mg) were measured in an aluminum container under a dry nitrogen flow at a heating or cooling rate of 10 °C/min. α -Alumina was used as a standard.

The Computer-Generated Molecular Models of the Titanium Compounds

Molecular mechanics were carried out using the CAChe system (CAChe Scientific) to optimize the titanium compounds. MM2 parameters were used as force field parameters, and optimization was done by conjugate gradient method until the convergence energy became less than 0.001 kcal/mol. The molecular model with the lowest energy was determined by generating an exhaustive potential energy map computed from dihedral search labels for all the single bonds searching between -180° and 120° in 5 steps and all the double bonds searching between -180° and 90° in 3 steps, followed by picking up 10 conformers from the one with the lowest energy and further optimization of them.

REFERENCES AND NOTES

 For recent reviews, see: (a) Ziegler Catalysts; Fink G., Mülhaupt R., Brinzinger H. H., Eds.; Springer-Verlag: Berlin, 1995. (b) Brinzinger, H. H.; Fischer, D.; Mülhaupt, R.; Rieger, B.; Waymouth, R. M. Angew. Chem. Int. Ed. Engl. 1995, 34, 1143.

- (2) For a recent review, see: Hatada, K.; Kitayama, T. In Macromolecular Design of Polymeric Materials; Hatada, K., Kitayama, T., Vogl, O., Eds.; Marcel Dekker: New York, 1997; p. 139.
- (3) For recent reviews, see: (a) Sawamoto, M. Prog. Polym. Sci. 1991, 16, 111.
 (b) Kennedy, J.P.; Iván, B. Designed Polymers by Carbocationic macromolecular Engineering: Theory and Practice; Hanser: Munich, 1992.
 (c) Cationic Polymerizations; Matyjaszewski, K., Ed.; Marcel Dukker: New York, 1996.
- (4) Higashimura, T.; Yonezawa, T.; Okamura, S.; Fukui, K. J. Polym. Sci. 1959, 39, 487.
- (5) Kunitake, T.; Aso, C. J. Polym. Sci., A-1 1970, 8, 665.
- (6) Schildknecht, C.E.; Zoss, A. O.; McKinley, C. Ind. Eng. Chem. 1947, 39, 180.
- (7) Natta, G.; Dall'Asta, G.; Mazzanti, G.; Giannini, U.; Cesca, S. Angew. Chem.
 1959, 71, 205.
- (8) Yuki, H.; Hatada, K.; Ota, K.; Kinoshita, I.; Murahashi, S.; Ono, K.; Ito, Y. J. Polym. Sci., A-1 1969, 7, 1517.
- (9) Ohgi, H.; Sato, T. Macromolecules 1993, 26, 559.
- (10) (a) Sikkema, D. J.; Angad-Gaur, H. Makromol. Chem. 1980, 181, 2259. (b)
 Angad-Gaur, H.; Sikkema, D. J. Makromol. Chem. 1980, 181, 2385.
- (11) Okamura, S.; Higashimura, T.; Watanabe, T. Makromol. Chem. 1961, 50, 137.
- (12) Kamigaito, M.; Sawamoto, M.; Higashimura, T. Macromolecules 1995, 28, 5671.
- (13) Sawamoto, M.; Kamigaito, M. Macromol. Symp. 1996, 107, 43.
- (14) Hatada, K.; Kitayama, T.; Matsuo, N.; Yuki, H. Polym. J. 1983, 15, 719.
- (15) Kamigaito, M.; Maeda, Y.; Sawamoto M.; Higashimura, T. Macromolecules 1993, 26, 1643.

Effects of Monomer Structure on Stereospecificity

ABSTRACT

Stereoregulation in cationic polymerization of various alkyl vinyl ethers was investigated with bis[(2,6-diisopropyl)phenoxy]titanium dichloride [TiCl₂{OC₆H₄(*i*-Pr)₂(2,6-)}₂] (1; catalyst) in conjunction with the HCl-adduct of isobutyl vinyl ether (IBVE) as an initiator in *n*-hexane at -78 °C. The tacticities depend on the substituents of the monomers. Isobutyl and isopropyl vinyl ethers gave highly isotactic polymers (*mm* = 83%), whereas *tert*-butyl and *n*-butyl vinyl ethers resulted in lower isotactic contents (*mm* ~ 50%) similar to those with TiCl₄, a conventional Lewis acid, thus indicating that the steric bulkiness of the substituents is not the critical factor in stereoregulation. The statistical analysis revealed that the high isospecificty is achieved not by the chain end but by the catalyst 1 or the counteranion derived therefrom.

INTRODUCTION

The synthesis of stereoregular polymers has been one of the most important and attractive objects in polymer chemistry, and near perfect stereoregularity has been achieved in coordination¹ and anionic² polymerizations. In cationic polymerization, however, such fine stereoregulation has been difficult in homogeneous systems, partly because of the lack of designed catalysts or counteranions that can control the stereochemistry of carbocation.

Despite these backgrounds, as shown in Chapter 1, the author has found that a designed titanium Lewis acid, bis[(2,6-diisopropyl)phenoxy]titanium dichloride [TiCl₂{ $OC_6H_4(i-Pr)_2(2,6-)$ }₂ (1), in conjunction with the HCl adduct of isobutyl vinyl ether (HCl–IBVE) (initiator) induces isospecific cationic polymerization of IBVE ($m \ge 90\%$; *m* is the content of *meso* dyad) (Scheme 1).³ Similar 2,6-disubstituted phenoxy derivatives also led to fairly high (~70–80%) isotacticity but cannot exceed the value for 1. Therefore, it has not been revealed why only this titanium compound gives highly isotactic poly(IBVE) among various titanium compounds.

The steric structure of polymers is, in general, determined not only by catalysts but also by the structure of the growing carbocation (i.e., monomers). In cationic polymerization, for example, the tacticity of poly(alkyl vinyl ethers) depends upon the substituent or the alkyl group of the monomers and increases in the following order: alkyl; *tert*-butyl > isobutyl > methyl > isopropyl > ethyl.⁴ On the other hand, Kunitake et al. reported that the size of the counterion (catalyst) or the tightness of the growing ion pair is a critical factor.⁵

In this chapter, the author has therefore investigated the cationic polymerization of various alkyl vinyl ethers with 1 in conjunction with IBVE-HCl in *n*-hexane at -78 °C and discussed the effects of monomer substituent on the tacticity (Scheme 1). Additionally, a mechanism for the isospecific cationic

Effects of Monomer



Scheme 1

polymerization is discussed on the basis of a statistical analysis of their tacticity data.

Results and Discussion

1. Cationic Polymerization of Various Alkyl VEs with 1. Cationic polymerization of various alkyl VEs [alkyl = n-C₄H₉ (NBVE), t-C₄H₉ (TBVE), i-C₃H₇ (IPVE) n-C₃H₇ (NPVE), C₂H₅ (EVE), CH₃ (MVE)] was conducted with 1 in conjunction with the HCl–IBVE as an initiator in the presence of 2,6-di-*tert*-butyl-4-methylpyridine (DTBMP) in n-hexane at -78 °C, where the highly isotactic poly(IBVE) is obtained.³

As shown in Figure 1, the polymerization rates with 1/IBVE-HCl system depended on the monomers.⁶ For example, the VEs with a branched substituent (IPVE, TBVE, and IBVE) were polymerized much faster than those with a linear substituent (NPVE and NBVE). EVE was much less active, and no polymers formed from MVE under the same conditions. These results are probably due to that when the monomer's alkyl group is less bulky, the catalyst 1 can interact (form

-37-



Figure 1. Time-conversion curves for the polymerizations of various alkylVEs with IBVE-HCl/1/DTBMP in *n*-hexane at -78 °C: [M]₀ = 0.38 M; [IBVE-HCl]₀ = 5.0 mM; [1]₀ = 10 mM; [DTBMP]₀ = 2.0 mM. VE: (O) IPVE; (**①**); TBVE; (**●**) IBVE; (**①**) NBVE; (**△**) NPVE; (**▲**) EVE; (**■**) MVE.

a complex) more easily and thus less available for the dissociation of the terminal C-Cl bond into the growing cation. Similar results were also obtained with TiCl₄ in place of **1**, where no polymers were formed even from EVE (see Table 1 entry 11). Such reduced rates are also due to the strong interaction between the highly oxophilic titanium metal and oxygen of VE. These results suggest that the Lewis acid catalysts can also interact with the oxygen attached to the substituent and that such interaction may affect the stereoselectivity of the growing species as described below.

2. Steric Structure of Poly(VE)s. The tacticity of the poly(VE)s thus obtained was measured by ¹³C NMR spectroscopy in C_6D_6/CCl_4 (9/1 v/v) at 55 °C. Regardless of the pendant alkyl groups, the spectra gave well-resolved main-chain methine or methylene signals that are split due to the steric structures (the asterisked carbons in Figure 2).⁴ For example, the dyad tacticity of poly(IBVE) was determined from the main-chain methylene carbons [$-CH_2$ -CH(Oi-Bu)-] around 39-42 ppm and the pendant methylene carbons [$-OCH_2$ -CH(CH_3)₂] around 76-77 ppm (the spectrum not shown) gave the corresponding triad data. For the other poly(alkyl VE)s, the tacticity was determined similarly. Table 1 summarizes these tacticity data along with the number-average molecular weights (M_n) and the molecular weight distribution (M_w/M_n).



Figure 2. ¹³C NMR spectra [67.9 or 125.8 MHz, CDCl₃ or CCl₄/C₆D₆ (9/1 v/v), 55 °C] of poly(VE)s obtained with IBVE-HCl/1/DTBMP in *n*-hexane at -78 °C: see Figure 1 for reaction conditions.

				time,	conv,	$M_{ m n}$		-	triad, 9	%	dyac	l, %
ntry	monomer	R	MX_n	min	%	X 10 ⁻⁴	$M_{\rm W}/M_{ m n}$	шш	mr	rr	ш	r
	IBVE	<i>i-</i> Bu	1	5	76	3.65	2.01	83	10	٢	606	10
7	IBVE	<i>i</i> -Bu	TiCl4	10	94	3.78	1.92	46	40	14	73	26
ŝ	NBVE	<i>n</i> -Bu		17	95	2.58	2.72	56	35	6	76	24
4	NBVE	n-Bu	TiCl4	8400	59	3.78	1.92	45	43	12	69	31
S.	TBVE	<i>t</i> -Bu	Ţ		93	5.76	2.19	49	41	10	<i>q</i> 69	31b
9	TBVE	t-Bu	TiCl4	1	86	4.72	2.11	46	40	14	66b	34b
7	IPVE	<i>i</i> -Pr	ł		98	11.0	3.96	83	11	9	889	12b
8	IPVE	<i>i</i> -Pr	TiCl4	0.33	88	9.72	2.18	69	20	11	62	21
6	NPVE	n-Pr	1	25	91	3.30	2.06	0	J	с Г	78	22
10	EVE	Et	***	1620	78	1.48	3.80	2	g		64	26
11	EVE	Et	TiCl4				lod ou	ymn				

 $[IBVE-HCI]_0 = 5.0 \text{ mM}; [TiCl_4]_0 = 2.0 \text{ mM} \text{ in } n\text{-hexane at } -78 \text{ °C}.$ bCalculated from the triad values: m = mm+ (1/2)mr; r = rr + (1/2)mr.⁸ cNot determined. a

Chapter 2

-40-

(a) Effects of Pendant Alkyl Groups. The tacticities of the poly(VE)s clearly depended on the alkyl groups in the monomers. IBVE and IPVE, the pendant groups of which are branched near the ether oxygen, gave relatively high isotacticity (m = 88-90%, Table 1; entry 1 and 7). In contrast, VE with linear (NBVE, NPVE, and EVE) or highly bulky (TBVE) substituents gave less isotactic contents (Table 1; entry 3, 9, 10, and 5). Especially, the poly(EVE) and poly(TBVE) had lower isotacticity (m = 64 and 69%, respectively). The tacticity is thus determined not by the stereo bulkiness but the structure (or branching) of the alkyl substituents, in contrast to the results for the conventional cationic polymerization with BF3OEt2, where the bulkiest t-butyl derivative (TBVE) gave the highest isotacticity.^{8,9} When combined with our previous results with different Lewis acids,³ the observed trend implies that the factor determining the tacticity with the 1/HCl-IBVE initiating system is not only the bulkiness of the counteranion from Lewis acid but also the structure of the pendant groups, or the interaction between the Lewis acid and the pendant ether oxygen of VE.

(b) Effects of Lewis Acids (Catalysts). Some of the samples showed 13 C NMR signals of the carbons adjacent to the ether oxygen, from which triad tacticity can be determined, and these resonances depended on the Lewis acids, as well as the alkyl groups. Figure 3 compares these peaks for poly(IBVE), poly(NBVE), poly(TBVE), and poly(IPVE) samples obtained with 1 and TiCl4 (see also Table 1). In all four systems, 1 led to higher isotacticity than TiCl4 did. In particular, the difference is distinct (83% vs 46%) in poly(IBVE) and poly(IPVE). Though less pronounced, a similar difference is found for poly(NBVE) and poly(TBVE). Note that, with TiCl4, isotacticity is almost independent for the those butyl polymers, in contrast to the clear dependence with bulkier 1 as a catalyst. Such a high dependence suggests some interaction between 1 and the substituents. As for poly(IPVE), a relative high isotacticity was obtained with TiCl4, though still lower than that with 1 (69% vs 83%).

-41-



Figure 3. ¹³C NMR spectra [67.9 or 125.8 MHz, CDCl₃ or CCl₄/C₆D₆ (9/1 v/v), 55 °C] of poly(VE)s obtained with IBVE–HCl/1/DTBMP or IBVE–HCl/TiCl₄ in *n*-hexane at -78 °C: (A), (C), (E), and (G) $[M]_0 = 0.38$ M; [IBVE–HCl]_0 = 5.0 mM; [1]_0 = 10 mM; [DTBMP]_0 = 2.0 mM, (B), (D), (F), and (H) $[M]_0 = 0.38$ M; [IBVE–HCl]_0 = 2.0 mM.

(c) Tacticity Statistics. To elucidate the polymerization mechanism or the transition state of the growing carbocation, the author further examined the statistics of the triad distributions. First, if he assumes the chain end model, namely the one-parameter model proposed by Bovey, the triad distribution [I (isotactic; mm), H (heterotactic; mr), S (syndiotactic; rr)] is expressed as:¹⁰

$$I = \sigma_1^2$$
(1)

$$H = 2(\sigma_1 - \sigma_1^2)$$
(2)

$$S = (1 - \sigma_1)^2$$
(3)

where the parameter σ_1 is the probability that a monomer adds a polymer chain to give the isotactic (dyad meso) configuration as that of the preceding unit. In

Figure 4, the probability (P, %) of each tacticity is plotted against σ_1 obtained by eq 1. The solid lines are theoretical (the Bovey plot, eq 1-3) for the chain-end control. The plots of H and S for NBVE/TiCl₄ (\blacksquare) and TBVE/TiCl₄ (\blacklozenge), which gave the lower isotaciticity (Table 1), seem to follow this model. In contrast, the plots for IBVE/1 (O), and IPVE/1 (Δ), and IPVE/TiCl₄ (\blacklozenge), which gave the higher I value, clearly deviate from the calculated lines. This suggests that the stereo chemistry of these systems follows some statistics other than chain-end control.



Figure 4. Relationship between σ₁ and triad tacticity of poly(VE)s with IBVE-HCl/1/DTBMP or IBVE-HCl/TiCl₄ in n-hexane at -78 °C;
(O) IBVE/1; (●) IBVE/TiCl₄; (△) IPVE/1; (▲) IPVE/TiCl₄; (□) NBVE/1; (■) NBVE/TiCl₄; (◇) TBVE/1; (◆) TBVE/TiCl₄. The curves indicate the theoretical values calculated from the Bovey's model.

Chapter 2

On the other hand, for a system if tacticity is controlled by the catalyst; i.e., the so-called enantiomorphic-site control, the following equations hold:

$$I = 1 - 3(\sigma_2 - \sigma_2^{2})$$
(4)

$$H = 2(\sigma_2 - \sigma_2^{2})$$
(5)

$$S = \sigma_2 - \sigma_2^{2}$$
(6)

where the parameter σ_2 represents the persistence probability that a monomer adds a polymer chain in either configuration determined by the catalyst (or the



Figure 5. Relationship between σ₂ and triad tacticity of poly(VE)s with IBVE-HCl/1/DTBMP or IBVE-HCl/TiCl₄ in *n*-hexane at -78 °C;
(O) IBVE/1; (●) IBVE/TiCl₄; (△) IPVE/1; (▲) IPVE/TiCl₄; (□) NBVE/1; (■) NBVE/TiCl₄; (◇) TBVE/1; (◆) TBVE/TiCl₄. The curves indicate the theoretical values calculated from the enantiomorphic catalyst sites model.

counteranion therefrom).¹¹ Figure 5 shows the plots for the observed probability (P, %) against σ_2 as defined by eq 4. The solid lines are drawn according to the equations 4–6. In contrast to Figure 4, the plots of H and S for IBVE/1 (O), IPVE/1 (Δ), and IPVE/1 (Δ) agree well with the calculated lines, whereas the plots for NBVE and TBVE seem inconsistent with the calculated lines.

These results show that the high isospecificity $(m \sim 90\%)$ in the polymerizations of IBVE and IPVE with IBVE-HCl/1 is due to not the chain-end control but the catalyst control, most probably originated from 1 or the counteranion therefrom. Although the enantiomorphic site control has already been applied to the cationic polymerization of methyl vinyl ether with heterogeneous catalysts, there have been no examples for homogeneous systems even for isospecific cationic polymerization of BF₃OEt₂.¹² However, 1 is not effective for all the alkyl vinyl ethers for example, the TBVE system seems to follow the chain-end control (\diamond in Figure 4) even with the use of 1, probably due to the bulky *t*-butyl substituent. The polymerization of NBVE with TiCl₄ is controlled by chain end (\blacksquare in Figure 4), but one with 1 can involve the enantiomorphic-site (catalyst) control (\square in Figure 5). Thus, the Lewis acid, 1 can effectively control the tacticity in the cationic polymerization of alkyl vinyl ethers, although the controllability also depends on the substituents of monomer.

Considering these results, the author proposes a new transition state [(A) in Scheme 2] for the IBVE–HCl/1 system, where the Lewis acid interacts with the anion from the initiator chloride and most probably with the ether oxygen of the incoming monomer. When this monomer-counteranion interaction operates, 1 can effectively control the direction of the incoming monomer's approach to the growing cation to result in stereoregulation. Other transition states for conventional cationic polymerization have been proposed elsewhere [(B) and (C) in Scheme 2],¹³ which do not involve interaction between the counteranion and the incoming monomer. Apparently, these two transition states cannot be applied for our systems with high isospecificity.



Scheme 2. Transition Sate for Growing Carbocation

In conclusion, 1 proved effective in isospecific cationic polymerization of IBVE and IPVE in conjunction with IBVE-HCl, to give relatively high isotactic content (mm = 83% or $m \sim 90\%$) that has not been attained by other cationic catalysts. The author proposed that this is most probably due to the effective interaction between 1 and the incoming monomer via the enantiomorphic-site control.

EXPERIMENTAL

Materials

IBVE, NBVE, EVE (Tokyo Kasei; purity >99%), TBVE (Aldrich; 99%), IPVE, and NPVE (Nippon Carbide Ind. 99%) were washed with 10% aqueous sodium hydroxide and then with water, dried overnight over potassium hydroxide (pellets), and distilled twice from calcium hydride before use. The initiator (HCl–IBVE)¹⁴ and the catalyst, TiCl₂[OC₆H₃*i*-Pr₂(2,6-)]₂ (1),³ were prepared as reported. Toluene (Wako; >99%) as solvent was dried overnight over calcium chloride, distilled from sodium benzophenone ketyl. Carbon tetrachloride (internal standard for gas chromatography; >99%) were dried overnight over

-46-

calcium chloride, distilled twice from calcium hydride. TiCl₄ (Aldrich; >99%) and DTBMP (Aldrich; 98%) was used as received.

Polymerization Procedures

Polymerization was carried out under dry nitrogen in baked glass tubes equipped with a three-way stopcock. A typical example for the polymerization procedure of IBVE is given below: The reaction was initiated by sequential addition of prechilled solutions of IBVE–HCl (in *n*-hexane; 0.30 mL) and mixture of 1 and DTBMP (in toluene, 0.30 mL) via dry syringes into a monomer solution (in toluene; 2.4 mL) containing IBVE (0.15 mL) and CCl₄ (0.15 mL). After predetermined intervals, the polymerization was terminated with prechilled methanol (2.0 mL) containing a small amount of ammonia. Monomer conversion was determined from its residual concentration measured by gas chromatography with CCl₄ as an internal standard. The polymer yield by gravimetry was in good agreement with the gas-chromatographic conversion of the monomer.

The quenched reaction mixture was washed with dilute hydrochloric acid, aqueous sodium hydroxide solution, and then with water to remove initiator residues, evaporated to dryness under reduced pressure, and vacuum dried to give the product polymers.

Measurements

The MWD of the polymers were measured by size-exclusion chromatography (SEC) in chloroform at 40 °C on three polystyrene gel columns [Shodex K-805L (pore size: 20–1000 Å; 8.0 nm i.d. × 30 cm) × 3; flow rate 1.0 mL/min] that were connected to a Jasco PU–980 precision pump and a Jasco 930-RI refractive index detector. The columns were calibrated against 11 standard polystyrene samples (Pressure Chemicals; $M_n = 580-1547000$; $M_w/M_n \le 1.1$) as well as the styrene monomer. ¹H NMR and ¹³C NMR spectra of the produced polymers were recorded in CDCl₃ or CCl₄/C₆D₆ (9/1 v/v) at 55 °C on a JEOL

-47-

LNM-LA500 or a JEOL JNM-GSX270 spectrometer, operating at 500.2 or 270.2 MHz (¹H) and 125.8 or 67.9 MHz (¹³C), respectively.

REFERENCES AND NOTES

- For recent reviews, see: (a) Ziegler Catalysts; Fink. G., Mülhaupt R., Brinzinger H. H., Eds.; Springer-Verlag: Berlin, 1995. (b) Brinzinger, H. H.; Fischer, D.; Mülhaupt, R.; Rieger, B.; Waymouth. R. M. Angew. Chem. Int. Ed. Engl. 1995, 34, 1143. (c) Coates, G. W. Chem. Rev. 2000, 100, 1223.
- (2) For a recent review, see: Hatada, K.; Kitayama, T. In Macromolecular Design of Polymeric Materials; Hatada, K., Kitayama, T., Vogl, O., Eds.; Marcel Dekker: New York, 1997; p. 139.
- (3) Ouchi, M.; Kamigaito, M.; Sawamoto, M. Macromolecules 1999, 32, 6407.
- (4) Hatada, K.; Kitayama, T.; Matsuo, N.; Yuki, H. Polym. J. 1983, 15, 719.
- (5) Kunitake, T.; Aso, C. J. Polym. Sci., A-1 1970, 8, 665.
- (6) Yuki, H.; Hatada, K.; Nagata, K.; Emura, T. Polym. J. 1970, 1, 269.
- (7) Frisch, H. L.; Mallows, C. L.; Bovey, F. A. J. Chem. Phys. 1966, 45, 1565.
- (8) Ohgi, H.; Sato, T. Macromolecules 1993, 26, 559.
- (9) Ohgi, H.; Sato, T. Macromolecules 1999, 32, 2403.
- (10) Bovey, F. A.; Tiers, G. V. D. J. Polym. Sci. 1960, 44, 173.
- (11) Shelden, R.A.; Fueno, T.; Tsunetsugu, T.; Furukawa, J. Polym. Lett. 1965, 3, 23.
- (12) Higashimura, T.; Ohsumi, Y.; Kuroda, K.; Okamura, S. J. Polym. Sci., A-1 1967, 5, 863.
- (13) Higashimura, T.; Masuda, T.; Okamura, S.; Yonezawa, T. J. Polym. Sci., A-1 1969, 7, 3129.

(14) Katayama, H.; Kamigaito, M.; Sawamoto, M. Macromolecules 1995, 28, 3747.

Chapter 3

Design of Counteranions with Bulky Phosphoric Acids as Initiato**rs**

ABSTRACT

In cationic polymerization of isobutyl vinyl ether (IBVE) with binary initiating systems consisting of a protonic acid as an initiator and a Lewis acid as an activator/catalyst, phosphoric acid derivatives [(RO)₂POOH] coupled with SnCl₄ gave highly isotactic poly(IBVE)s, where those with a bulky substituent (R), [C₄H₉CH(C₂H₅)CH₂O]₂POOH (7) and $(n-C_{10}H_{21})_2$ POOH (8), led to the highest isotacticity [*meso* dyad (*m*) = 86%]. In contrast, isospecificity was lower with IBVE-HCl and CF₃COOH under the same conditions. From the effects of polymerization temperature (-78 to 0 °C), it was concluded that the high isospecificity with 7 and 8 is due to an enthalpic factor.

INTRODUCTION

For high stereoregulation in cationic polymerization, which has been difficult, 1-5 the author has recently employed a binary system⁶ consisting of a protonic acid (HB) as an initiator and a Lewis acid (MX_n) as an activator.⁷ Obviously, not only stability or reactivity but also the stereochemistry of the growing species can be affected by steric and electronic interaction with the counteranion (BMX_n^{-}) to be generated from the anionic part (B^{-}) of HB and the catalyst MX_n . This implies that design of the initiator HB or the catalyst MX_n (or both) might lead to stereospecific cationic polymerization. For example, the combination of a bulky Lewis acid, bis[(2,6-diisopropyl)phenoxy]titanium dichloride [TiCl₂{ $OC_6H_4(i-Pr)_2(2,6-)$ }, with the HCl-adduct of isobutyl vinyl ether (HCl-IBVE; 2) leads to highly isotacitc poly(IBVE) ($m \ge 90\%$; m = mesodyad) (Scheme 1).⁷ Alternatively, aromatic acetals with sterically hindered substituents as initiators also give fairly high isospecificity ($m \sim 81\%$).⁸ Similar approaches have been reported by Aoshima^{9a} and Hashimoto.^{9b} who employed hindered esters (as additives; m = 62-66%) and hindered carboxylic acids (initiators; m = 66-69%), respectively.

From these backgrounds, the author focused our attention on the tacticity control by designed bulky counteranions derived from initiators (HB or B⁻) for



Scheme 1

IBVE cationic polymerization (Scheme 2). This approach is considered complementary with the previous one, as presented in Chapter 1, via sterically designed Lewis acids [TiCl₂(OAr)₂; Ar = substituted phenyl]. The protonic acids include HCl,¹⁰ CF₃COOH,¹¹ and a series of phosphoric acids [(RO)₂POOH] with bulky substituents R (4–8).¹² The author reports herein new highly isospecific cationic polymerization of IBVE with selected (RO)₂POOH (7 and 8).



Scheme 2

Results and Discussion

1. Various Combinations of HB and MX_n . (a) Effects of Acid Anions (B⁻). IBVE was polymerized in toluene at -78 °C with three protonic acids (initiators) 2-4 carrying simple but different anions in conjunction with various MX_n , ZnCl₂, SnCl₄, EtAlCl₂, TiCl₄, and 1; in the latter, the first four are conventional metal halides for cationic polymerization, whereas 4 is a designed Chapter 3

Lewis acid to give high isospecificity with Cl⁻ anion (m > 90%).^{7a} The results are summarized in Table 1.

entry	MX _n	HB	time	conv,	M _n	$M_{\rm w}/M_{\rm n}$	m (dyad),
			··· ···	%			%
1	ZnCl ₂	2	1.5 h	87	6,300	1.33	70
2		3	5 days	83	7,900	1.71	69
3		4	4 days	97	3,400	1.68	69
4	SnCl ₄	2	1 min	99	8,900	1.11	74
5		3	1 min	100	9,500	5.21	75
6		4	1 min	97	24,100	2.03	80
7	EtAlCl ₂	2	5 min	100	21,000	2.41	76
8		3	1 min	100	34,200	2.01	78
. 9		4	25 sec	97	55,000	1.83	78
10	TiCl ₄	2	1 min	93	13,900	1.66	72
11		3	30 sec	100	1,4000	2.15	79
12		4	5 min	99	1,0000	2.49	79
13	1	2	1 h	87	96,900	3.05	82
14		3	30 sec	9 8	88,400	4.06	77
15		4	20 h	86	457,000	1.93	76

Table 1. Polymerization of IBVE by HB/MX $_n^a$

^{*a*} [M]₀ = 0.38 M; [HB]₀ = 5.0 mM; [MX_n]₀ = 5.0 mM (10 mM only for ZnCl₂); in toluene at -78 °C.

In addition to variation of polymerization rate and polymer molecular weight, the stereoregularity of the products depended on the combination of HB and MX_n. Figure 1 shows some ¹³C NMR spectra of the main-chain methylene



Figure 1.

¹³C NMR spectra [125.8 MHz, CCl₄/C₆D₆ (9/1 v/v), 55 °C] around 39–42 ppm of poly(IBVE)s obtained with HB/MX_n in toluene at -78 °C: [M]₀ = 0.38 M; [HB]₀ = 5.0 mM; [MX_n]₀ = 5.0 and 10 (only for ZnCl₂) mM.

region $[-CH_2-CH(Oi-Bu)-]$ for poly(IBVE)s obtained with some specific combination, and the meso (*m*) and the racemo (*r*) dyad contents were calculated from the integration ratios (see also Table 1).¹³ Although all these polymers are more or less isotactic, the signal patterns and the isospecificity (*m*) therefrom clearly depended on the HB/MX_n combinations. For example, ZnCl₂ led to a lower isotacitic content (*m* ~ 70%) irrespective of the acids anions (Table 1, entry 1–3; Figure 1A–1C), probably due to the smaller size of the counteranion. With SnCl₄, TiCl₄, and **4**, the meso content was higher and dependent on the initiators (Table 1, entry 4–6, 10–12, and 13–15; Figure 1D–11). Thus, the tacticity of poly(IBVE) obtained with the binary system was influenced with not only Lewis acid but also the anionic part of the initiator. In particular, a phosphoric acid derivative, **4**, gave a high isotacticity (*m* = 80%) in conjunction with SnCl₄, which

Chapter 3

is comparable to that for the 2/1 pair (m = 82%; entry 13 in Table1). Namely, a bulky protonic acid (4) seems as effective as a bulky Lewis acid (1) for high isospecificity. However, the combination of bulky two components, 4 (initiator) and 1 (MX_n) resulted in a lower *m* content (m = 76%; entry 15 in Table1). Thus, bulkiness is not a single factor for stereoregulation and some shape or structural factor should also be considered, and phosphoric acids seem interesting initiators.

(b) Effects of SnCl₄ Concentration. A factor critical in the stereoregulation in ionic polymerization is the ionic dissociation of the growing carbocation,¹⁴ and this can be modulated not only by the structural variation of the cations (monomers) and the counteranions (initiating systems) but also by the concentration of the carbocationic species. For the HB/MX_n systems, the latter factor is in part controlled by the concentration of the MX_n catalyst. Thus, effects

		[SnCl ₄] ₀ ,		conv,			<i>m</i> (dyad),
entry	HB	mM	time	%	M _n	$M_{\rm w}/M_{\rm n}$	%
1	2	10	25 sec	99	7,100	1.25	71
2		5.0	1 min	99	8,900	1.11	74
3		1.0	3 min	94	8,700	1.11	. 77
4	3	10	25 sec	99	6,200	1.61	72
5		5.0	1 min	100	9,500	5.21	75
6		1.0	2 h	57	10,000	3.95	79
7	4	10	25 sec	98	11,000	1.40	77
8		5.0	1 min	97	24,100	2.03	80
9		1.0	2 h	93	15,000	2.06	84

 Table 2. Effects of SnCl₄ Concentration on Cationic Polymerization of

 IBVE with HB/SnCl₄^a

a [M]₀ = 0.38 M; [HB]₀ = 5.0 mM in toluene at -78 °C.

-56-

of SnCl₄ concentration were examined on polymerization with the three initiators (2-4), where the [SnCl₄]₀ was varied in the range 1.0-10 mM (Table 2). The number-average molecular weights (M_n) and molecular weight distributions (MWDs) were clearly dependent on the combination and the concentration. Controlled M_n and narrow MWDs were obtained with 1 irrespective of [SnCl₄]₀. On the other hand, 2 and 3 resulted in less controlled molecular weights and MWDs even at low [SnCl₄]₀, probably due to slow initiation and interconversion between the dormant and the active species.

Figure 2 shows ¹³C NMR spectra of the poly(IBVE)s obtained at 10 and 1.0 mM of SnCl₄ (see also Figure 1D–F for 5.0 mM SnCl₄). With all initiators, the lower [SnCl₄]₀ is, the stronger the meso signals, and therefore the *m* values monotonously increased at lower catalyst concentration (Figure 3). Similar effects were observed in the cationic polymerization of *tert*-butyl VE with H₂O/BF₃OEt₂ system.¹⁵ Because the lower [SnCl₄]₀ concentration means



Figure 2. ¹³C NMR spectra [125.8 MHz, CCl_4/C_6D_6 (9/1 v/v), 55 °C] around 39–43 ppm of poly(IBVE)s obtained with HB/SnCl₄ in toluene at -78 °C: [M]₀ = 0.38 M; [HB]₀ = 5.0 mM; [SnCl₄]₀ = 10 and 1.0 mM.

dilution of a weak electrolyte solution, which in turn facilitates the ionic dissociation of the growing cation, the enhanced isospecificity seems inconsistent with the notion that less dissociated or ion pair species is favorable for isotactic propagation. The author suspects that the rather complicated action of SnCl₄ upon interaction of cationogen (initiator like 2) might be responsible for this opposite trend in stereoregulation (see below).¹⁶



Figure 3. Relationship between $[SnCl_4]_0$ and meso content (m, %) of poly(IBVE) with HB/SnCl₄ in toluene at -78 °C: $[M]_0 = 0.38$ M; $[HB]_0 = 5.0$ mM; $[SnCl_4]_0 = 10, 5.0$ and 1.0 mM.

Perhaps more important, the phosphoric acid 3 proved more effective in affording isotactic poly(IBVE) than 1 and 2, to give 84% meso dyad at the lowest SnCl₄ concentration.

2. Phosphoric Acid/SnCl₄ Systems: Isospecific Bulky Anions. (a) Polymerizations. The formation of iso-rich poly(IBVE) with 4/SnCl₄ system promoted him to investigate a series of phosphoric acids [(RO)₂POOH] (4-8) with bulky substituents as initiators in conjunction with SnCl₄ at a low concentration (1.0 mM) (Scheme 1).

Figure 4 shows time-conversion curves of the polymerizations and the MWDs of the obtained poly(IBVE)s. All the polymerizations proceeded quantitatively, and the rates were dependent on the initiators, in contrast to that with similar systems coupled with $ZnCl_2$ at -15 °C, where they were almost independent of the substituents.¹² This may be due to an anion exchange reaction between the acid and SnCl₄, as considered later.



Figure 4. Time-conversion curves and MWD curves for the polymerizations of IBVE with (RO)₂POOH/SnCl₄ in toluene at -78 °C: $[M]_0 =$ 0.38 M; $[(RO)_2POOH]_0 = 5.0$ mM; $[SnCl_4]_0 = 1.0$ mM. (RO)₂POOH: (**①**) 5; (**①**) 6; (**▲**) 7; (**●**) 4; (**♦**) 8.



Figure 5. ¹³C NMR spectra [125.8 MHz, CCl_4/C_6D_6 (9/1 v/v), 55 °C] around 39–43 ppm of poly(IBVE)s obtained with (RO)₂POOH/SnCl₄ in toluene at -78 °C: [M]₀ = 0.38 M; [(RO)₂POOH]₀ = 5.0 mM; [SnCl₄]₀ = 1.0 mM. The spectrum (F) with **2/1** is from ref 8a.

(b) Stereoregularity. The meso dyad of the poly(IBVE)s with 5–8 were examined by ¹³C NMR (Figure 5). All of those initiators gave high isotacticity, among which the phosphates with a long and bulky alkyl group (7 and 8) induced the highest isotacticity (m = 86%; Figure 5D and E). These isotactic polymers formed hard films similar to those obtained with the bulky Lewis acid 1 (m = 90%; see also Figure 5F).^{7a}

	temperature,			conv,			m (dyad),	
entry	HB		°C	time	%	M _n	$M_{\rm w}/M_{\rm n}$	%
1		ſ	-78	2 h	98	27,900	3.04	86
2			-60	30 sec	98	26,800	2.39	81
3	7	{	-40	1 min	99	32,700	2.64	78
4			-20	3 min	99	14,000	2.64	74
5		l	0	10 min	92	6,600	2.50	70
6		ſ	78	3 min	94	8,700	1.11	77
7			-60	1 min	99	9,300	1.08	76
8	2	$\left\{ \right.$	-40	1 min	100	8,700	1.11	73
9			-20	1 min	100	7,400	1.43	69
10		l	0	1 min	100	7,300	3.74	68

 Table 3. Effects of Temperature on Cationic Polymerization of

 IBVE with HB/SnCL4

 $\overline{a \, [M]_0} = 0.38 \, M; \, [HB]_0 = 5.0 \, mM; \, [SnCl_n]_0 = 1.0 \, mM$ in toluene.

(c) Effects of Polymerization Temperature. Finally, effects of polymerization temperature were examined in the polymerization with 7 coupled with SnCl₄; the same set of experiments were carried out with the chloride 2 for comparison (Table 3). Unexpectedly, polymerization rate with the 7/SnCl₄ system showed a complicated temperature dependence: -60 °C ~ -40 °C > -20 °C

> 0 °C > -78 °C. In contrast, the polymerization rate with $2/SnCl_4$ simply increased at higher temperature.

Figure 6 shows the temperature dependence of the poly(IBVE) tacticity, according to the Fordham equation:¹⁷

$$\ln(m/r) = (\Delta Sm^{\ddagger} - \Delta Sr^{\ddagger})/R - (\Delta Hm^{\ddagger} - \Delta Hr^{\ddagger})/RT$$

where $\Delta S_m^{\ddagger} - \Delta S_r^{\ddagger}$ is the difference in activation entropy and $\Delta H_m^{\ddagger} - \Delta H_r^{\ddagger}$ is that in activation enthalpy between isotactic (*m*) and syndiotactic (*r*) enchainment, *R* is the gas constant (8.3145 J/mol·K), and *T* is the absolute temperature. In both systems, a good linear relationship holds between the $\ln(m/r)$ and the reciprocal of the temperature, whereas the slope and the intercept were quite different between



Figure 6.

Relationship between 1/T and $\ln(m/r)$ of poly(IBVE)s obtained with 2/SnCl₄ and 7/SnCl₄ in toluene at -78, -60, -40, -20, or 0 °C: [M]₀ = 0.38 M; [HB]₀ = 5.0 mM; [SnCl₄]₀ = 1.0 mM. System: 2/SnCl₄ (\blacktriangle); 7/SnCl₄ (\bigcirc).

Design of Initiators

the two systems. The phosphoric acid system obviously involves a high enthalpy difference, consistent with its high isospecificity.

The high isoselectivity by the phosphoric acids is probably due to exchange reaction with SnCl₄ to generate a bulkier Lewis acid e.g., SnCl₃[OP(O)(OR)₂], as previously reported for similar systems.¹² The tin metal has higher coordination number, and hence a bulkier counteranion can be formed to induce isospecificity. The new bulkier Lewis acid might further interact with the phosphate-capped dormant end ~~~C–OP(O)(OR)₂ to form a bulkier counteranion carrying two bulky phosphates. In contrast, with the use of a series of acetic acids as an initiator, such an exchange reaction does not occur and resulted in lower isospecificity.¹⁸

In Part I, the author demonstrates an effective isospecificity enchainment by a bulky protonic acid and/or a bulky Lewis acid from which the counteranion forms.

EXPERIMENTAL

Materials

IBVE (Tokyo Kasei; >99%) was washed with 10% aqueous sodium hydroxide and then with water, dried overnight over potassium hydroxide (pellets), and distilled twice from calcium hydride before use. Toluene (Wako; >99%) were dried overnight over calcium chloride, distilled from sodium benzophenone ketyl. Carbon tetrachloride (internal standard for gas chromatography; >99%) were dried overnight over calcium chloride, distilled twice from calcium hydride. The protonic acids (initiators) and the Lewis acids except for 1 were obtained commercially and used without further purification: CF₃COOH (3; Wako; >99%), (*n*-BuO)₂POOH (4; Tokyo Kasei; >97%), (PhO)₂POOH (5; Aldrich; 99%), Chapter 3

(PhCH₂O)₂POOH (6; Tokyo Kasei; >98%), $[C_4H_9CH(C_2H_5)CH_2O]_2POOH$ (7; Tokyo Kasei; >97%), $(n-C_{10}H_{21})_2POOH$ (8; Tokyo Kasei; >95%), ZnCl₂ (Aldrich; 1.0 M solution in Et₂O), SnCl₄ (Aldrich; 1.0 M solution in CH₂Cl₂), EtAlCl₂ (Kanto Chemicals; 1.0 M solution in *n*-hexane), TiCl₄ (Aldrich; >99%). The titanium compound 1⁷ and the HCl–IBVE adduct 2¹⁹ were synthesized as already reported.

Polymerization Procedures

Polymerization was carried out under dry nitrogen in baked glass tubes equipped with a three-way stopcock. A typical example for the polymerization procedure is given below: The reaction was initiated by sequential addition of prechilled solutions of HB (0.015 mmol; 0.30 mL of 0.050 M in toluene) and MX_n (0.015 mmol; 0.30 mL of 0.050 M in toluene) via dry syringes into a monomer solution (in toluene; 2.4 mL) containing IBVE (1.14 mmol; 0.15 mL) and CCl4 (0.15 mL). The total volume of the reaction mixture was thus 3.0 mL. After predetermined intervals, the polymerization was terminated with prechilled methanol (2.0 mL) containing a small amount of ammonia. Monomer conversion was determined from its residual concentration measured by gas chromatography with CCl₄ as an internal standard. The polymer yield by gravimetry was in good agreement with the gas-chromatographic conversion of the monomer.

The quenched reaction mixture was washed with dilute hydrochloric acid, aqueous sodium hydroxide solution, and then with water to remove initiator residues, evaporated to dryness under reduced pressure, and vacuum dried to give the product polymers.

Measurements

The MWD of the polymers were measured by size-exclusion chromatography (SEC) in chloroform at 40 °C on three polystyrene gel columns [Shodex K-805L (pore size: 20–1000 Å; 8.0 nm i.d. \times 30 cm) \times 3; flow rate 1.0

-64-

mL/min] that were connected to a Jasco PU-980 precision pump and a Jasco 930-RI refractive index detector. The columns were calibrated against 11 standard polystyrene samples (Pressure Chemicals; $M_n = 580-1547000$; $M_w/M_n \le 1.1$) as well as the styrene monomer. ¹³C NMR spectra of the produced polymers were recorded in CCl₄/C₆D₆ (9/1 v/v) at 55 °C on a JEOL LNM-LA500 spectrometer, operating at 500.2 MHz (¹H) and 125.8 MHz (¹³C), respectively.

REFERENCES AND NOTES

- (1) Schildknecht, C. E.; Zoss, A. O.; McKinley, C. Ind. Eng. Chem. 1947, 39, 180.
- (2) Yuki, H.; Hatada, K.; Ota, K.; Kinoshita, I.; Murahashi, S.; Ono, K.; Ito, Y. J. Polym. Sci., A-1 1969, 7, 1517.
- (3) Ohgi, H.; Sato, T. Macromolecules 1993, 26, 559.
- (4) Natta, G.; Dall'Asta, G.; Mazzanti, G.; Giannini, U.; Cesca, S. Angew. Chem. 1959, 71, 205.
- (5) (a) Sikkema, D. J.; Angad-Gaur, H. Makromol. Chem. 1980, 181, 2259. (b)
 Angad-Gaur, H.; Sikkema, D. J. Makromol. Chem. 1980, 181, 2385.
- (6) For recent reviews on cationic polymerization, see: (a) Sawamoto, M. Prog. Polym. Sci. 1991, 16, 111. (b) Kennedy, J. P.; Iván, B. Designed Polymers by Carbocationic Macromolecular Engineering: Theory and Practice; Hanser: Munich, 1992. (c) Matyjaszewski, K., Ed.; Cationic Polymerizations, Marcel Dekker: NY, 1996.
- (7) (a) Chapter 1 of this thesis; Ouchi, M.; Kamigaito, M.; Sawamoto, M. *Macromolecules* 1999, 32, 6407. (b) Chapter 2 of this thesis; Ouchi, M.; Kamigaito, M.; Sawamoto, M. J. Polym. Sci. Part A, in press.
- (8) Satoh, K.; Kamigaito, M.; Sawamoto, M. Macromolecules 2000, 33, 748.

- (9) (a) Aoshima, S.; Ito, Y.; Kobayashi, E. Polym. J. 1993, 25, 1161. (b) Hashimoto, T.; Iwata, T.; Uchiyama, N.; Kodaira, T. J. Polym. Sci. Part A: Polym. Chem. 1999, 37, 2923.
- (10) (a) Kamigaito, M.; Maeda, Y.; Sawamoto, M.; Higashimura, T. *Macromolecules* 1993, 26, 1643. (b) Kamigaito, M.; Sawamoto, M.; Higashimura, T. *Macromolecules* 1995, 28, 5671.
- (11) Kamigaito, M.; Sawamoto, M.; Higashimura, T. Macromolecules 1991, 24, 3988.
- (12) (a) Sawamoto, M.; Kamigaito, M.; Higashimura, T. Polym. Bull. 1988, 20, 407. (b) Kamigaito, M.; Sawamoto, M.; Higashimura, T. J. Polym. Sci. Part A: Polym. Chem. 1993, 31, 2987.
- (13) Hatada, K.; Kitayama, T.; Matsuo, N.; Yuki, H. Polym. J. 1983, 15, 719.
- (14) Kunitake, T.; Aso, C. J. Polym.Sci., A-1 1970, 8, 665.
- (15) Ohgi, H.; Sato, T. Macromolecules 1999, 32, 2403.
- (16) Katayama, H.; Kamigaito, M.; Sawamoto, M. Macromolecules 1998, 31, 4703.
- (17) Fordham, J. W. L.; J. Polym. Sci. 1959, 39, 321.
- (18) Kamigaito, M.; Sawamoto, M.; Higashimura, T. Macromolecules 1992, 25, 2587.
- (19) Katayama, H.; Kamigaito, M.; Sawamoto, M. Macromolecules 1995, 28, 3747.

-66-

PART II

PRECISION CATIONIC POLYMERIZATION OF CYCLOPENTADIENE: CONTROL OF MOLECULAR WEIGHT AND REGIOSPECIFICITY

Design of Initiating Systems for Control of Molecular Weight of Poly(CPD)

ABSTRACT

Precise control of molecular weights and their distributions (MWD) in cationic polymerization of cyclopentadiene (CPD) has first been achieved with three-component initiating system, the HCl-adduct of CPD or a vinyl ether (initiator), SnCl₄ (Lewis acid as catalyst), and *n*-Bu₄NCl (additive), in dichrolomethane at -78 °C. The number-average molecular weights of the polymers increased in direct proportion to monomer conversion, as well as to the monomer-to-initiator mole ratio (up to 200), and the MWDs were narrow throughout the reactions ($M_w/M_n = 1.2-1.3$), clearly narrower than those in the additive-free systems ($M_w/M_n > 2$). Weak bases such as diethyl ether and ethyl acetate also proved effective as additives, whereas THF inhibited CPD polymerization. The poly(CPD) carried the initiator fragment at the α -end (nearly one per chain), and consisted of nearly the same contents of the 1,2- and 1,4-repeat units; free from other enchainments.

INTRODUCTION

Hydrocarbon polymers with rigid cyclic repeat units have attracted much attention as new materials of attractive thermal, mechanical, and electronic properties such as high glass-transition temperature and low dielectric constant. A majority of such polymers have been obtained via addition (vinyl) polymerization of cyclic olefins,^{1,2} cyclopolymerization of α , ω -dienes,² and ring-opening metathesis polymerization of bicyclic olefins.³ Among the three methods, the first gives the most rigid backbones where cyclic units are directly connected each other without spacers, whereas the second and the third methods generate polymers with one and two spacer carbons, respectively, in the main chain. The monomers for the addition polymerization include cyclic dienes, which afford rigid main chains of directly linked, unsaturated rings that may be hydrogenated into more stable saturated structures.⁴

Cyclopentadiene (CPD) is a representative cyclic diene that is easily obtained by the retro Diels-Alder reaction of dicyclopentadiene, and can be polymerized via cationic addition mechanism with conventional Lewis acids (metal halides) and stable carbocation salts (eq 1).⁵ The growing species therein is most likely a relatively stable allylic carbocation and thus leads to 1,2- and 1,4enchainments in the main-chain cyclic units. Despite its long history of research dating back to the 1920s,^{6–10} the precise control of CPD cationic polymerization has not been attained, in terms of molecular weights, main-chain microstructure (1,2- vs. 1,4-); occasional formation of insoluble products renders research more difficult. The poor controllability is also seen in the transition metal-catalyzed coordination polymerizations developed more recently.^{11,12}



-70-
One of the most effective techniques to control polymer molecular weight and structures is living polymerization. Living cationic polymerization¹³ has been achieved for various vinyl monomers, such as vinyl ethers, styrene derivatives, isobutene, indene, *N*-vinylcarbazole, and β -pinene, where the obtained polymers have controlled molecular weights and narrow molecular weight distributions (MWDs).¹⁴ Most of living cationic polymerizations can be achieved by binary system that consists of a protonic acid (HB) or its derivative as an initiator and a Lewis acid (MX_n) as an activator (eq 2). The Lewis acid activates the C-B bond in the initiator and polymer terminals to give a growing cationic species (~~C+) accompanied by a counteranion (BMX_n⁻). The choice of initiators, Lewis acids, and additives is crucial in achievement of living polymerization.¹⁴



Given these backgrounds, the author became interested in precision control of CPD cationic polymerization for at least two reasons: There are few or no living cationic polymerizations of linear and cyclic aliphatic dienes; no general methods are available for controlling the 1,2- and 1,4-structures in the CPD main chain, despite that the selective formation of such units would be important in the chemistry of allylic-type growing carbocations and would also give interesting materials (see above). This study was thus to develop living cationic polymerization of CPD with HB/Lewis acid initiating systems; the effects of additives and solvent polarity were also examined for optimization of reaction control. This chapter reports the first example, to his knowledge, precise control of polymer molecular weights and MWDs in cationic CPD polymerization, with use of the HCl–CPD adduct (1)/SnCl4 initiating system in the presence of an additive such as n-Bu4NCl, diethyl ether, and ethyl acetate.

Chapter 4



RESULTS AND DISCUSSIONS

1. Search of Initiators

As potential initiators for living cationic polymerizations of CPD, we first examined hydrogen chloride adducts of CPD (1), 2-chloroethyl vinyl ether (2), α -



Figure 1. Time-conversion curves for the polymerization of CPD with R-Cl/SnCl₄ in toluene at -78 °C: $[M]_0 = 0.50 \text{ M}$; $[R-Cl]_0 = 5.0 \text{ mM}$; $[SnCl_4]_0 = 50 \text{ mM}$. R-Cl: (\bigcirc) 1; (\blacktriangle) 2; (\blacksquare) 3; (\bigcirc) 4.

methylstyrene (3), and styrene (4). The use of these adducts means that, in eq 2, the anion B⁻ is invariably the chloride, while the carbocationic parts differ in structure and reactivity. The polymerizations were carried out in toluene at -78°C with SnCl₄ as the activator (MX_n) in eq 2: [CPD]₀/[Initiator]₀/[SnCl₄]₀ = 500/5.0/50 mM. All systems induced polymerization, whereas the rate depended upon the initiators (Figure 1). The CPD adduct 1 induced a fast and quantitative polymerization. The polymerizations with 2 and 3 proceeded at nearly the same rate as with 1 at the initial stage but slowed down subsequently. In contrast, 4 resulted in a much slower and incomplete polymerization.



Figure 2. MWD curves of poly(CPD) obtained with R-Cl/SnCl₄ in toluene at -78 °C: $[M]_0 = 0.50 \text{ M}$; $[R-Cl]_0 = 5.0 \text{ mM}$; $[SnCl_4]_0 = 50 \text{ mM}$. R-Cl: (A) 1; (B) 2; (C) 3; (D) 4.

Figure 2 shows the MWDs of the obtained polymers, as measured by sizeexclusion chromatography (SEC) with a polystyrene calibration. The SEC curves with 1 were relatively narrow and shifted to higher molecular weights with increasing CPD conversion. The adducts 2 and 3 also gave relatively narrow MWDs until ~50% conversion, however, they became broader at higher conversion. The MWDs with 4 were much broader due to slow initiation from 4, which has a stronger and less active C-Cl bond than the others. Thus, 1 proved most suitable for controlling cationic polymerization of CPD with SnCl₄.

2. Effects of Solvents

In general, cationic polymerization is affected by solvent polarity and the effects of solvents were then examined in the CPD polymerization with $1/SnCl_4$ at -78 °C. As shown in Figure 3, the higher the polarity of solvent is, the faster the



Figure 3. Time-conversion and MWD curves for the polymerization of CPD with $1/SnCl_4$ at -78 °C: $[M]_0 = 0.50$ M; $[1]_0 = 5.0$ mM; $[SnCl_4]_0 = 5.0$ or 50 (only for MCH) mM. Solvent: (\blacktriangle) CH₂Cl₂; (O) toluene/CH₂Cl₂ (3/1 v/v); ($\textcircled{\bullet}$) toluene; (\blacksquare) MCH.

rate of polymerization. Addition of CH_2Cl_2 to toluene (toluene/ $CH_2Cl_2 = 3/1$ v/v) made the polymerization faster, where conversion reached 96% in 15 min. The polymerization in CH₂Cl₂ was very fast and completed within 1 min to give products of multimodal SEC curves. On the other hand, the polymerization in methylcyclohexane (MCH), a less polar solvent, was much slower even with 10 times more $SnCl_4$ (50 mM) than in CH_2Cl_2 . These results show that the polymerization of CPD highly depends on solvent polarity. The MWDs of the polymers were fairly narrow in toluene or its mixture with CH₂Cl₂. The multimodal MWD in CH₂Cl₂ is probably due to a highly active, dissociated growing carbocation, which can attack a double bond in another polymer chain to generate comb-like and partially cross-linked polymers. In contrast to the polymerization with SnCl₄ alone in CH₂Cl₂ reported previously,⁸ no insoluble gels formed with the 1/SnCl₄ system. Overall, therefore, less polar media are better suited for control of CPD polymerization.

3. Effects of Additives: Control of CPD Polymerization

Control of polymerization in polar media like CH₂Cl₂ may be possible by addition of an ammonium salt or a base, as demonstrated in cationic polymerizations of vinyl ethers^{15–17}, styrene derivatives¹⁸, and isobutene^{19,20}. These additives are expected to affect the nature of the growing carbocation by suppressing ionic dissociation or stabilizing the growing carbocation. As shown in Figure 4, addition of a chloride, *n*-Bu₄NCl, retarded the polymerization in CH₂Cl₂ and gave poly(CPD)s with much narrower MWDs ($M_w/M_n \sim 1.2$) than those in the salt free systems.

The number-average molecular weights (M_n) of the polymers increased in direct proportion to monomer conversion and agreed well with the calculated values assuming that one polymer chain forms per CPD-HCl adduct molecule, although they were based on a polystyrene calibration (Figure 5). Therefore, *n*-Bu₄NCl proved an effective additive for control of molecular weights and MWDs

--75--



Figure 4. Effects of added *n*-Bu₄NCl on CPD polymerization with 1/SnCl₄ in CH₂Cl₂ at -78 °C: [CPD]₀ = 500 mM; [1]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu₄NCl]₀ = 5.0 mM. Additive: (**O**) none; (**•**) *n*-Bu₄NCl.





 $M_{\rm n}, M_{\rm w}/M_{\rm n}$, and MWD curves of poly(CPD) obtained with 1 (\bullet , \blacktriangle) or 2 (O, \triangle)/SnCl₄ in the presence of *n*-Bu₄NCl in CH₂Cl₂ at -78 °C: [CPD]₀ = 500 mM; [1 or 2]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu₄NCl]₀ = 5.0 mM. The diagonal bold line indicates the calculated $M_{\rm n}$ assuming the formation of one polymer chain per 1 molecule.

in cationic polymerization of CPD. Another initiator 2, also gave polymers with controlled molecular weights and narrow MWDs in conjunction with n-Bu₄NCl. To his knowledge, these are the first examples of control molecular weights and MWDs of poly(CPD).

A series of oxygen-based nucleophiles such as diethyl ether (Et₂O), isopropyl ether (*i*-Pr₂O), ethyl acetate (EA), and tetrahydrofuran (THF) were also employed for the control of the polymerization with $1/\text{SnCl}_4$ in CH₂Cl₂ at -78 °C (Figure 6).^{16,17} Addition of Et₂O and EA retarded the polymerization moderately, where the conversion reached 95% in 2 h and 93% in 16 h, respectively. With both additives, the MWDs were as narrow as those obtained with *n*-Bu₄NCl. The M_n increased linearly with conversion and agreed with the



Figure 6. Time-conversion and MWD curves of poly(CPD) obtained with 1/SnCl₄ in the presence of additives in CH₂Cl₂ at -78 °C: [CPD]₀ = 500 mM; [1]₀ = 5.0 mM; [SnCl₄]₀ = 10 mM, Additive 1vol% ([Et₂O]₀ = 96 mM, [EA]₀ = 100 mM, [*i*-Pr₂O]₀ = 71 mM, [THF]₀ = 120 mM). Additive: (**O**) Et₂O; (**●**) EA; (**♦**) *i*-Pr₂O; (**■**) THF.

calculated values. The polymerization with i-Pr₂O, a weaker and hindered base, was fast and uncontrolled, similar to that in the absence of additives. In contrast, no polymers were obtained with THF, a stronger base. Thus, the 1/SnCl4 initiating system afforded poly(CPD) with controlled molecular weight and narrow MWDs in the presence of bases such as n-Bu₄NCl, Et₂O, and EA.

Figure 7 is to demonstrate further control of molecular weights and the distributions by the ratio of monomer to initiator $([M]_0/[Initiator]_0)$. Thus, in the $1/SnCl_4/n$ -Bu₄NCl initiating system, the amount of 1 was fixed ([1]_0 = 5.0 mM), whereas the initial monomer concentration was varied ([M]_0 = 250, 500, and 1000



Figure 7. M_n , M_w/M_n , and MWD curves of poly(CPD) obtained with 1/SnCl₄ in the presence of *n*-Bu₄NCl in CH₂Cl₂ at -78 °C: [CPD]₀ = 1000 (\bullet), 500 (\bullet), 250 (\blacktriangle) mM; [1]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu₄NCl]₀ = 5.0 mM. The diagonal bold line indicates the calculated M_n assuming the formation of one polymer chain per 1 molecule.

mM; $DP_n = 50$, 100, and 200, respectively) (Figure 7). The M_n increased linearly to monomer conversion and agreed well with the respective calculated values. These results also show that 1 in fact serves as an initiator to form polymer with controlled molecular weights.

The living nature of the polymerization with the $1 / \text{SnCl}_4$ initiating system in the presence of *n*-Bu₄NCl or Et₂O was then examined in monomer-addition experiments; thus CPD was polymerized with these systems and a fresh feed of CPD was added to the reaction mixture when the initial charge had almost been consumed. Although a second-phase polymerization proceeded, it was slower than the first and terminated at 50–70% conversion of the added CPD, to give products of a bimodal SEC curve with a small shoulder of lower molecular weights. This indicates that a part of the growing carbocations reacts with the double bond of a polymer chain, not that of monomer, to be inactivated at the latter stage of the first-phase polymerization where most of the monomer has been consumed. Thus, the cationic polymerizations of CPD with these systems can not be defined as "living polymerization", despite that the observed molecular weight increase and the narrow MWDs show a long lifetime of the intermediate and the near absence of side-reactions.

5. Polymer Analysis

The structure of the poly(CPD) obtained with $2/\text{SnCl}_4/n$ -Bu₄NCl in CH₂Cl₂ was examined by ¹H NMR spectroscopy (Figure 8). The vinyl ether adduct 2 was used herein for easy detection of the initiator fragment. The poly(CPD) shows large characteristic signals of main chain repeat units, i.e., olefinic (*e*) and aliphatic protons (*f*-*i*). The aliphatic protons consist of roughly four parts due to 1,2- and 1,4-structures, as known in conventional cationic polymerization of CPD.²¹ The 1,2-unit content was 46% based on peak areas $[(1 - f/e) \times 100]$.²¹ The stereoselectivity was almost the same in both salt-present and salt-free systems. This point is discussed in Chapter 5 in detail.



Figure 8. ¹H NMR spectrum (500 MHz, CDCl₃) of poly(CPD) ($M_n = 2900$, $M_w/M_n = 1.15$) obtained with 2/SnCl₄ in the presence of *n*-Bu₄NCl in CH₂Cl₂ at -78 °C: [CPD]₀ = 500 mM; [2]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu₄NCl]₀ = 5.0 mM.

In addition to these major peaks, there appeared minor absorptions assignable to the initiator fragment (a-d). The M_n (NMR) was determined from the relative peak area of the main-chain repeat units (e) to the initiator moiety (b-d), and was very close to that by SEC $[M_n(NMR) = 3000, M_n(SEC) = 2900]$. This also suggests that 2 serves as the initiator that forms one living polymer chain per molecule.

The poly(CPD) obtained with 1/SnCl₄/n-Bu₄NCl and 2/SnCl₄/n-Bu₄NCl was further analyzed by matrix-assisted laser-desorption-ionization time-of-flight mass spectroscopy (MALDI-TOF-MS) (Figure 9). Both spectra consist of a series of sharp peaks separated by a 66 Da interval each, corresponding to the molecular weight of CPD monomer. These indicate the virtual absence of

-80-



Figure 9. MALDI-TOF-MS spectrum of poly(CPD) obtained with 1/SnCl₄ (A) and 2/SnCl₄ (B) in the presence of *n*-Bu₄NCl in CH₂Cl₂ at -78 °C: [CPD]₀ = 500 mM; [1 or 2]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu₄NCl]₀ = 5.0 mM. (A): $M_n = 3700$, $M_w/M_n = 1.14$ (B): $M_n = 2900$, $M_w/M_n = 1.15$.

significant side reactions such as chain-transfer. Furthermore, there is 41 Da difference between the sharp peaks in the two spectra, which agrees with the mass difference between 1 and 2. This result also indicates that one initiator generates one polymer chain and that every polymer chains contains the initiator moiety at the α -end. The terminal structure predicted from the absolute mass was olefin structure, which is due to chlorine elimination induced by laser similar to the Cl-terminated polymers obtained with transition metal-catalyzed living radical polymerization or due to hydrogen chloride elimination during the work-up process.²²

In conclusion, the author has demonstrated the synthesis of poly(CPD) with controlled molecular weights and narrow MWDs via cationic polymerization of

-81-

CPD with the 1 or 2/SnCl₄ initiating system in the presence of *n*-Bu₄NCl, Et₂O or EA in CH₂Cl₂ at -78 °C. To his knowledge, this is the first example of precise control of molecular weights and the distributions in cationic polymerization of CPD.

EXPERIMENTAL

Materials

Unless otherwise specified, all of the chemicals were purchased from Wako company, Japan. CPD was obtained by the retro Diels-Alder reaction of dicyclopentadiene (Tokyo Kasei; >95%) at 165 °C over calcium hydride and distilled from calcium hydride. It was stored at -84 °C and used within one The purity and the absence of dicyclopentadiene in CPD was confirmed month. by NMR before use (purity >99%). CH₂Cl₂ (>99%) was dried overnight over calcium chloride, distilled from phosphorous pentoxide and then from calcium hydride before use. Methylcyclohexane (MCH) (Tokyo Kasei; >99%), carbon tetrachloride (internal standard for gas chromatography; >99%), ethyl acetate (EA) (>99%) and 1-phenylethyl chloride (4; >97%) were dried overnight over calcium chloride, distilled twice from calcium hydride. Toluene (>99%), diethyl ether (Et₂O) (>99%), and tetrahydrofuran (THF) (>99%) were dried overnight over calcium chloride, distilled from sodium benzophenone ketyl. SnCl4 (Aldrich; 1.0 M solution in CH₂Cl₂), *n*-Bu₄NCl (Tokyo Kasei; >98%) and isopropyl ether (*i*-Pr₂O) (Aldrich; anhydrous; 99%) were used as received.

Synthesis of HCl Adduct 1–3

The HCl adducts 1-3 were synthesized by bubbling dry HCl gas into a solution of the respective vinyl compounds (CPD, CEVE, and α MeSt).²³ The

-82-

synthesis of HCl–CPD adduct 1 as a typical example was done below. Dry HCl gas was bubbled into 1.0 M toluene solution (36 mL) of CPD (3.0 mL; 0.036 mol) at -78 °C. The gas was generated by dropping concentrated sulfuric acid into powdery sodium chloride, and dried by passing it through a column packed with calcium chloride. After 30 min, the excess HCl in the reaction mixture was removed by bubbling dry nitrogen gas at 0 °C for 30 min, and the clean and quantitative formation of adduct 1 was confirmed by ¹H NMR spectroscopy. The concentration of 1 was determined by titrating the chloride by the Volhard method. The observed value (0.89 M) further confirmed the quantitative formation of 1 from CPD.

Polymerization Procedures

Polymerization was carried out under dry nitrogen in baked glass tubes equipped with a three-way stopcock. A typical example for the polymerization procedure in CH₂Cl₂ is given below: The reaction was initiated by sequential addition of prechilled solutions of 1 (0.02 mmol; 0.40 mL of 0.050 M in CH₂Cl₂) and SnCl₄ (0.02 mmol; 0.40 mL of 0.050 M in CH₂Cl₂) via dry syringes into a monomer solution (in CH₂Cl₂; 3.2 mL) containing CPD (2.0 mmol; 0.165 mL) and CCl₄ (0.165 mL). The total volume of the reaction mixture was thus 4.0 mL. For polymerization in the presence of additives, these were dissolved in SnCl₄ solution prior to initiation. After predetermined intervals, the polymerization was terminated with prechilled methanol (2.0 mL) containing a small amount of ammonia. 3-tert-Butyl-4-hydroxy-5-methylphenyl sulfide (0.014 mmol; 0.005 g) was then added as an antioxidant immediately after quenching reaction. Monomer conversion was determined from its residual concentration measured by gas chromatography with CCl₄ as an internal standard. The polymer yield by gravimetry was in good agreement with the gas-chromatographic conversion of the The guenched reaction mixture was washed with dilute hydrochloric monomer.

acid and with water to remove initiator residues, evaporated to dryness under reduced pressure, and vacuum dried to give the product polymers.

Measurements

MWD of the polymers were measured by size-exclusion The chromatography (SEC) in chloroform at 40 °C on three polystyrene gel columns [Shodex K-805L (pore size: 20–1000 Å; 8.0 nm i.d. \times 30 cm) \times 3; flow rate 1.0 mL/min] that were connected to a Jasco PU-980 precision pump and a Jasco 930-RI refractive index detector. The columns were calibrated against 11 standard polystyrene samples (Pressure Chemicals; $M_{\rm p} = 580-1547000$; $M_{\rm w}/M_{\rm p} \le 1.1$) as well as the styrene monomer. ¹H NMR spectra of the produced polymers were recorded in CDCl₃ on a JEOL LNM-LA500 spectrometer, operating at 500.2 MHz. MALDI-TOF-MS spectra were measured on a Perseptive Biosystems Voyager-DE STR equipped with a 2-m linear flight tubes and a 337-nm nitrogen laser (pulse width, 3 ns), along with a delayed extraction capability. All experiments were carried out at an accelerating potential of 20 kV. In general, mass spectra from 256 laser shots were accumulated summed to produce a final spectrum. Angiotensin I (human; MW = 1296.5) (BACHEM) and insulin (bovine pancreas 28.3; MW = 5733.50 (nacalai) were used as internal standards to calibrate the mass scale. Samples for MALDI-TOF-MS analysis were prepared by mixing the poly(CPD), a matrix (1,8-dihydroxy-9(10H)-anthracenone; dithranol), and a cationizing agent (silver trifluoroacetate) in THF. For example, a poly(CPD) sample (approximately 10 mg/mL) was dissolved in THF. The matrix and the cationizing agent were also dissolved separately in THF (30 mg/mL and 10 mg/mL, respectively). A 30 μ L portion of the poly(CPD) solution, 90 μ L of the matrix solution, and 30 μ L of the cationizing agent were mixed in a glass vial. Then 0.5 μ L portions of the mixture were deposited onto 10–20 wells of the gold-coated sample plate and dried in air at room temperature.

REFERENCES AND NOTES

- (1) Okuda, J.; Mülhaupt, R. In Synthesis of Polymers (Materials Science and Technology Series), Schlüter, A. –D., Ed.; Wiley-VCH, Weinheim, Germany, Chapter 5, 1999.
- (2) Coates, G. W. Chem. Rev. 2000, 100, 1223.
- (3) Grubbs, R. H.; Khosravi, E. In Synthesis of Polymers (Materials Science and Technology Series), Schlüter, A. –D., Ed.; Wiley-VCH, Weinheim, Germany, Chapter 3, 1999.
- (4) Recently, Natori et al. reported living anionic polymerization of 1,3-cyclohexadiene and synthesis of high T_g polymers by hydrogenation of the product polymers; (a) Natori, I. *Macromolecules* 1997, 30, 3696. (b) Natori, I.; Imaizumi, K.; Yamagishi, H.; Kazunori, M. J. Polym. Sci. Part B: Polym. Phys. 1998, 36, 1657.
- (5) For a review on cationic polymerization; see: Kennedy, J. P. Cationic Polymerization of Olefins: A Critical Inventory, Wiley-Interscuence, New York, 1975.
- (6) Staudinger, H.; Bruson, H. A. Ann. 1926, 447, 110.
- (7) Aso, C.; Kunitake, T.; Ishimoto, Y. J. Polym. Sci. Part A-1 1968, 6, 1163.
- (8) Sigwalt, P.; Vairon, J. P. Bull. Soc., Chim. France 1964, 482.
- (9) Momiyama, Z.; Imanishi, Y.; Higashimura, T. Chem. High Polymers, 1966, 23, 56.
- (10) Aso, C.; Kunitake, T.; Ishimoto, Y. J. Polym. Sci. Part A-1 1968, 6, 1175.
- (11) McCann, M.; Coda, E. M. G. J. Mol. Catal.: A. Chem. 1996, 109, 99.
- (12) Kühn, F. E.; Ismeier, J. R.; Schon, D.; Xue, W.-M.; Zhang, G.; Nuyken, O. Macromol. Rapid Commun. 1999, 20, 555.
- (13) Miyamoto, M.; Sawamoto, M; Higashimura, T. Macromolecules 1984, 17, 265.

-85-

- (14) For recent reviews on living cationic polymerization, see: (a) Sawamoto, M. Prog. Polym. Sci. 1991, 16, 111. (b) Kennedy, J. P.; Iván, B. Designed Polymers by Carbocationic Macromolecular Engineering: Theory and Practice; Hanser: Munich, 1992. (c) Matyjaszewski, K., Ed.; Cationic Polymerizations, Marcel Dekker: NY, 1996.
- (15) Kamigaito, M.; Maeda, Y.; Sawamoto, M.; Higashimura, T. Macromolecules 1993, 26, 1643.
- (16) Aoshima, S.; Higashimura, T. Polym. Bull. 1986, 15, 417.
- (17) Kishimoto, Y.; Aoshima, S.; Higashimura, T.Macromolecules 1989, 22, 3877.
- (18) Ishihama, Y; Sawamoto, M.; Higashimura, T. Polym. Bull. 1990, 24, 201.
- (19) Kaszas, G.; Puskas, J.; Kennedy, J. P. Polym. Bull. 1988, 20, 413.
- (20) Perneker, T.; Kennedy, J. P. Polym. Bull. 1991, 26, 305.
- (21) Aso, C.; Kunitake, K.; Ito, Y.; Ishimoto, Y. J. Polym. Sci., Part B: Polym. Lett. 1966, 4, 701.
- (22) Nonaka, H.; Ouchi, M.; Kamigaito, M.; Sawamoto, M. Macromolecules, in press.
- (23) Katayama, H.; Kamigaito, M.; Sawamoto, M. Macromolecules 1995, 28, 3747.

Design of Initiating Systems for Control of Regiospecificity in CPD Polymerization

ABSTRACT

The control of regioselectivity in cationic polymerization of cyclopentadiene (CPD) was examined with initiating systems consisting of HCl-adduct of CPD (1; initiator) and a Lewis acid (activator/catalyst), and among a series of metal halides as catalysts, weak Lewis acids such as ZnX_2 (X = Cl, Br, I) gave the highest [1,4]-contents (64, 70, 76%, respectively). In particular, the 1/ZnBr₂ system induced controlled polymerization to give poly(CPD)s with narrow molecular weight distributions (MWDs) ($M_w/M_n = 1.3-1.5$) and relatively high regiospecificity ([1,4] = 70%). In contrast, SnCl₄, TiCl₄, and other strong Lewis acids resulted in less controlled microstructures ([1,4] = 45-50%). Other reaction parameters, e.g., solvents, additives, and temperatures, did not dramatically affect regioselectivity, giving 55-60% 1,4-contents almost invariably.

INTRODUCTION

Cyclopentadiene (CPD) is one of the most representative and inexpensive cyclic dienes, and its polymer is a candidate for new hydrocarbon-polymeric materials with rigid cyclic repeat units. CPD can be polymerized by cationic catalysts to give two rigid main-chain cyclic units, 1,2- and 1,4-enchainments, due to the allylic growing carbocation.¹ Despite the long research history that dates back to 1920s,^{2–6} fine control of molecular weights and microstructures of the polymers is difficult, and the products thus far reported are invariably of low molecular weights and poorly controlled architectures.

As shown in Chapter 4, however, the author has succeeded in controlling molecular weights and molecular weight distributions (MWDs) in cationic polymerization of CPD by using an initiating system consisting of an HCl-adduct (1 or 2; initiator) and SnCl₄ (activator/catalyst) in the presence of an additive such as *n*-Bu₄NCl, diethyl ether, or ethyl acetate (eq 1).⁷ These polymerizations are considered to proceed via the allylic carbocation (3) reversibly generated from its dormant, chlorine-capped species, to give polymers with controlled molecular weights and narrow MWDs ($M_w/M_n = 1.2-1.3$).



-88-

However, even these systems failed to control another critical factor of CPD cationic polymerization, the regioselectivity for the main chain, giving nearly equal amounts of 1,2- and 1,4-units as do conventional systems.⁶ Another problem in cationic CPD polymerization is the formation of irregular main-chain structures (due to cation rearrangement) and insoluble products (due to the in-chain double bonds).^{2,6,8,9} Thus, the regiospecific cationic polymerization of CPD has not been attained yet, which is in contrast to that some metal-catalyzed coordination polymerizations of cyclopentene generate crystalline, highly isotactic *cis*-1,3-polymers.^{10–16}

Thus, this chapter is directed to develop regiospecific cationic polymerization of CPD with the binary initiating systems [1/MX_n (Lewis acid)], in relation to the effects of reaction conditions such as solvents, temperatures, additives, and Lewis acids (Scheme 1). As with other vinyl monomers,^{17–19} these reaction parameters may affect the nature of the growing CPD (allylic) carbocation and, in turn, the regiospecificity in propagation. For example, in cationic polymerization of isobutyl vinyl ether, the meso dyad contents (isotacticity) can be varied between 60



Solvent: toluene, CH_2Cl_2 , methylcyclohexane (MCH) Additive: *n*-Bu₄NCl, Et₂O, $CH_3CO_2Et_2$, *i*-Pr₂O, Ph₃CCl, (N, Y), (N, Y), (N, Y), (N, Y)MX_n: SnX₄ (X = Cl, Br), TiX₄ (X = Cl, Br), TiCl₂(O*i*-Pr)₂, AlBr₃, BCl₃, ZnX₂ (X = Cl, Br, I)

Scheme 1

and 90% with designed Lewis acids (see Chapter 1).¹⁹ None of such stereochemical control via catalyst design, however, are found in cationic polymerization of CPD and related cyclic or linear conjugated dienes.

In this chapter, the author reports that 1,4-enchainment can be varied between 51 and 76% with the use of various catalysts and that especially $1/\text{ZnBr}_2$ -system induces controlled cationic polymerization to give poly(CPD)s with controlled molecular weights, narrow MWDs ($M_w/M_n = 1.2-1.3$), and relatively high regiospecificity ([1,4] = 70%).

RESULTS AND DISCUSSIONS

1. Effects of Solvents, Additives, and Temperature.

(a) Solvents. In cationic polymerization of vinyl monomers, the steric structure of its polymers depends on solvent polarity, which affects the ionic dissociation of the growing carbocation.^{19–21} Therefore, CPD polymerization



Figure 1. Effects of solvents on the MWD curves of poly(CPD) obtained with 1/SnCl₄ at -78 °C: See Table 1 for reaction conditions.

--90--

was carried out with the $1/SnCl_4$ initiating system at -78 °C in various solvents, toluene, methylcyclohexane (MCH), and CH₂Cl₂ (Table 1, entry 1–3 and Figure 1, A–C). The higher the solvent polarity, the faster the polymerization, and the molecular weight and MWDs of the polymers depended on the solvents.

The microstructures of the obtained poly(CPD) was analyzed by ¹H NMR spectroscopy. Figure 2 shows the representative ¹H NMR spectrum of a sample obtained with $1/SnCl_4$ in toluene at -78 °C (for entry 3, Table1). Characteristic signals appear in two regions, one for the olefinic ptorons (D; $a^{1,2}$ and $a^{1,4}$) around 5.5–6.0 ppm and the other for the aliphatic ptorons ($b^{1,2}$, $c^{1,2}$, $d^{1,2}$, $b^{1,4}$, and $c^{1,4}$) around 1.5–3.0 ppm. The latters consist of well-resolved three parts; i.e., 1.4–1.8



Figure 2. ¹H NMR spectra (500 MHz, CDCl₃) of poly(CPD)s ($M_n = 7500$, $M_w/M_n = 1.62$, [1,2]/[1,4] = 45/55) obtained with 1/SnCl₄ in toluene at -78 °C: [CPD]₀ = 0.50 M; [1]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; entry 3, Table 1.

ppm (A; $c^{1,4}$), 1.8–2.2 ppm (B; $d^{1,2}$), and 2.2–2.8 ppm (C; $c^{1,2}$, $b^{1,2}$, and $b^{1,4}$). The 1,2- and 1,4-contents (hereafter, [1,2] and [1,4], respectively) can be calculated from their integrated signal intensities by the following two methods:³

> Method 1: [1,2], $\% = [(B+C-A)/(A+B+C)] \times 100$ Method 2: [1,2], $\% = [1-(A/D)] \times 100$

where A, B, C, and D represent the peak intensities of the four signals.

If a poly(CPD) sample consists of the two "regular" main-chain units, 1,2and 1,4-units, olefinic/aliphatic protons ratio D/(A+B+C) should be 0.5. If the ratio deviates from 0.5, the polymer contains other structures, such as those from the isomerization of the growing carbocation and saturated units via chain transfer or termination. The observed ratios for all of the poly(CPD) are close to 0.50, which indicates the absence of such side reactions under our reaction conditions with 1/SnCl₄. The regioselectivity did not depend on the solvents, either, where the [1,2] contents were around 45%.

(b) Additives. The effects of additives, which would interact with growing carbocation²² or Lewis acid activator,²³ were also examined with use of the 1/SnCl₄ system in CH₂Cl₂ at -78 °C (Table 1, entry 4–11 and Figure 3, A–H). In addition to *n*-Bu₄NCl,⁷ weak Lewis bases including diethyl ether, and ethyl acetate, Ph₃CCl and 2,6-di-tert-butyl-4-methylpyridine proved effective in controlling molecular weights and MWDs ($M_w/M_n = 1.3-1.4$) (Figure 3, A–E). However, diisopropyl ether, 2,4,6-trimethylpyridine, and 1-butylimidazole gave broader and multimodal MWDs (Figure 3, F–H), though some of them were effective in living cationic polymerization of styrene. In contrast, none of the additives affected the regiospecificity; i.e., all polymers showed about 45%–[1,2] and 55%–[1,4] contents (Table 1). Thus, control of stereospecificity in cationic polymerization with 1/SnCl₄ was difficult by changing solvent polarity or by adding salts and Lewis bases.

(c) Temperature. The regiospecificity was not affected by polymerization temperature (0 to -78 °C), either. Rather unusual in cationic polymerization, the

-92-

	I auto	I. Ellects u	IL DUIVEIIL AI	ninna nu		ru ron	merization wi	I I/DUCI	4 at -/8 (n)
	[SnCl4]0		[ad	lditive]0,	time,	conv,		[1,2]/	[1,4], %	- - - - -
entry	ШM	solvent	additive	ШM	min	%	$M_{ m n} \left(M_{ m W}/M_{ m n} ight)$	Meth. 1	Meth. 2	D/(A+B+C)
	5.0	toluene	none	0	120	91	7, 500 (1.62)	46 / 54	45/55	0.49
5	50	MCH^{b}	none	0	2940	47	4, 300 (2.17)	46 / 54	45 / 55	0.49
ŝ	5.0	CH ₂ Cl ₂	none	0	1	100	8, 600 (3.33)	47 / 53	45 / 55	0.48
4	5.0	CH ₂ Cl ₂	n-Bu4NCl	5.0	7	94	6, 500 (1.29)	46 / 54	46 / 54	0.50
5	10	CH ₂ Cl ₂	Et ₂ O	96	120	95	7, 100 (1.33)	44 / 56	45 / 55	0.51
9	10	CH ₂ Cl ₂	EA ^c 1	00	096	93	6, 600 (1.30)	46 / 54	46 / 54	0.51
٢	5.0	CH ₂ Cl ₂	Ph ₃ CCI	5.0	ŝ	87	5, 500 (1.29)	46 / 54	46 / 54	0.50
8	5.0	CH ₂ Cl ₂	DTBMPd	4.5	1	67	7, 700 (1.41)	48 / 52	47 / 53	0.50
6	10	CH ₂ Cl ₂	<i>i</i> -Pr ₂ O	71	1	66	6, 600 (3.67)	47 / 53	45 / 55	0.48
10	5.0	CH ₂ Cl ₂	TMPe	4.5	-	86	5, 800 (2.54)	48 / 52	47/53	0.49
11	5.0	CH ₂ Cl ₂	BIZf	4.5		100	8, 000 (2.44)	47 / 53	46 / 54	0.49
<i>a</i> [CPD]	0 = 500	mM; [1] ₀ =	= 5.0 mM; [S	$nCl_{4}]_{0} = 1$	5.0-501	nM; [A	dditive] $0 = 0-10$	00 mM.	b methylc	yclohexane.

Control of Regiospecificity

c ethyl acetate. d 2,6-di-tert-butyl-4-methylpyridine. e 2,4,6-trimethylpyridine. f 1-butylimidazole.

-93-



Figure 3. Effects of various additives on the MWD curves of poly(CPD) obtained with 1/SnCl₄ in CH₂Cl₂ at -78 °C: See Table 1 for reaction conditions.

overall polymerization rate with $1/SnCl_4$ in toluene did not clearly depend on temperature (conversion: 50% in 15 min; 80–100% in 120 min), though there were some differences in time-conversion profiles for the temperatures. The olefinic/aliphatic protons ratios was also close to 0.50; a slight decrease at 0 °C (47%) might indicate some side reaction at higher temperature, however the regioselectivity also unchanged, 55–60% [1,4] units.

2. Effects of Lewis Acids. Apart from external reaction parameters like solvent and temperature, the author employed various Lewis acids (MX_n) in conjunction with 1 in toluene or CH₂Cl₂ at -78 °C. The Lewis acids were: SnX₄ (X = Cl, Br), TiX₄ (X = Cl, Br), TiCl₂ $(Oi-Pr)_2$, AlBr₃, BCl₃, and ZnX₂ (X = Cl, Br, I). The results are summarized in detail in Table 2.

(a) Molecular Weight Control. Entry 1–5, Table 2, show the data for the polymerization in nonpolar toluene solvent. Strong Lewis acids such as TiX_4 (X = Cl, Br) and AlBr₃ induced rapid polymerizations (90% conversion within 10 min) to give broad MWDs. On the other hand, with a weak Lewis acid (BCl₃), a slower polymerization occurred (75% in 30 min, 100% in 5 h) and resulted in broader MWDs. Other weak Lewis acids such as SnBr₄, TiCl₂(O*i*-Pr)₂, and ZnX₂ (X = Cl, Br, I) were not active enough in toluene.

Marked effects of Lewis acids were observed in polar CH_2Cl_2 solvent (Figure 4 and entry 6–12, Table 2). Strong Lewis acids like TiX_4 (X = Cl, Br) and AlBr₃ gave insoluble gels due to crosslinking as previously reported for cationic polymerization with them in the absence of 1. In contrast, weaker Lewis acids such as SnBr₄, TiCl₂(O*i*-Pr)₂, and ZnX₂ (X = Cl, Br, I), which were not active in



Figure 4. MWD curves of poly(CPD) obtained with $1/MX_n$ in CH₂Cl₂ at -78 °C: [CPD]₀ = 0.50 M; [1]₀ = 5.0 mM; [MX_n]₀ = 5.0 mM or 100 mM (for ZnX₂).

		$[MX_n]_0,$			conv,		[1,2]/[1,4], %	
entry	MXn	Mm	solvent	time	%	$M_{\rm n} \left(M_{\rm W}/M_{\rm n} ight)$	Meth. 1	Meth. 2	D/(A+B+C)
-1	SnC14	5.0	toluene	2 h	91	7, 500 (1.62)	46 / 54	45 / 55	0.49
7	$TiCl_4b$	5.0	toluene	1 min	98	7, 700 (3.25)	44 / 56	43 / 57	0.49
ŝ	$TiBr_4b$	5.0	toluene	10 min	95	4, 700 (2.34)	49 / 51	48 / 52	0.49
4	$AlBr_3^{b,c}$	0.5	toluene	5 min	93	6, 700 (2.94)	45 / 55	43 / 57	0.49
S	BCl ₃	5.0	toluene	5 h	100	9, 200 (2.49)	42 / 58	41 / 59	0.50
9	SnC14	5.0	CH ₂ Cl ₂	1 min	100	8, 600 (3.33)	47 / 53	45 / 55	0.48
L	SnBr4	5.0	CH ₂ Cl ₂	15 min	70	4, 700 (24.4) <i>e</i>	47 / 53	46 / 54	0.50
8	TiCl ₂ (0 <i>i</i> -P ₁	r)2 5.0	CH ₂ Cl ₂	1 day	87	8, 700 (1.99)	41/59	41/59	0.50
6	BCl ₃	5.0	CH ₂ Cl ₂	4 min	LL	3, 100 (4.93) <i>e</i>	45 / 55	44 / 56	0.50
10	$ZnCl_2^d$	100	CH ₂ Cl ₂	26 days	63	4, 800 (1.37)	36 / 64	36 / 64	0.50
11	$ZnBr_2^d$	100	CH ₂ Cl ₂	5 days	92	8, 700 (1.37)	31./ 69	30 / 70	0.50
12	ZnI_2^d	100	CH ₂ Cl ₂	14 days	86	$11,200(15.3)^{e}$	26/74	24 / 76	0.48

-96-



Figure 5. M_n , M_w/M_n , and MWD curves of poly(CPD) obtained with $1/ZnBr_2$ in CH₂Cl₂ at -78 °C: [CPD]₀ = 0.50 M; [1]₀ = 5.0 mM; [ZnBr₂]₀ = 100 mM.

toluene, induced polymerizations in CH_2Cl_2 , despite the still low activity with some systems. Most of the Lewis acids gave bimodal or multimodal SEC curves, but $ZnCl_2$ and $ZnBr_2$ led to narrow MWDs that are similar to that for the controlled CPD polymers with $1/SnCl_4/n$ -Bu₄NCl system (Figure 3, trace A)⁷.

Especially with ZnBr₂, a quantitative polymerization occurred to give polymers with narrow MWDs ($M_w/M_n = 1.3-1.5$) (Figure 5). Also, the M_n increased in direct proportion to monomer conversion, as in the polymerization with $1/SnCl_4/n$ -Bu₄NCl in CH₂Cl₂ at -78 °C.⁷ Thus, $1/ZnBr_2$ proved effective in controlling molecular weights in cationic polymerization of CPD in a salt/additive free solvent.

(b) Regiospecificity with Lewis Acids. Table 2 also summarizes the 1,2and 1,4-microstructures of poly(CPD) samples obtained with a series of Lewis acids at -78 °C. The olefinic/aliphatic protons ratios [D/(A+B+C)] are nearly



Figure 6. ¹H NMR spectra (500 MHz, CDCl₃) around 1.3–3.2 ppm of poly(CPD) obtained with various Lewis acids coupled with 1 in CH₂Cl₂ at -78 °C. See Table 2 for reaction conditions.

independent of the catalysts and close to 0.50, as expected for the main chains exclusively with the regular 1,2- and 1,4-units.

In sharp contrast, Lewis acids clearly and markedly affected the regiospecificity ([1,4] content) in both toluene and in CH₂Cl₂. For example, Figure 6 shows the main-chain aliphatic proton regions of ¹H NMR spectra of poly(CPD) obtained in CH₂Cl₂ (see also Table 2, entry 6–12); a similar trend can be seen for the samples in toluene solvent (Table 2, entry 1–5). The 1,4-unit contents for CH₂Cl₂, determined from these spectra, are compared more visually in



Figure 7. 1,4-contents of poly(CPD) obtained under various conditions. See Table 1 and Table 2 for reaction conditions.

Figure 7 as a function of Lewis acids.

As already discussed, strong Lewis acids such as SnCl₄ and TiCl₄ give more or less conventional poor regioselectivity around 55%, irrespective of solvent polarity, additives, and molecular weight control (the so-called living nature of the relevant polymerizations). A similarly low 1,4-content (51%) can be found with CH₃SO₃H, a protonic acid sometimes employed for CPD cationic polymerization.²⁴

With weaker Lewis acids (the bottom 4 entries in Figure 6 and Table 2), importantly, the signals of the 1,2-units clearly weaken and, in turn, those of the 1,4-units increasingly become sharper and stronger in the order of $(SnCl_4 < TiCl_4 <)$ TiCl₂ $(Oi-Pr)_2 < ZnCl_2 < ZnBr_2 < ZnI_2$. In particular, ZnI₂ led to a 76% 1,4content, perhaps the highest, to his knowledge, reported thus far for cationically prepared poly(CPD). Another finding of interest is that ZnBr₂ gives a relatively high 1,4-content (70%), the second highest just below the iodide version, and that the bromide also allows a fine molecular weight control in salt-free CH₂Cl₂ solvent (see the previous section). Thus, given these dual controllability in both MWD and 1,4-content, the $1/ZnBr_2$ initiating system might lead to a regiospecific *and* living cationic polymerization of CPD, though optimum reaction conditions have not yet specified. The regiospecificity with weak Lewis acids is most probably due to formation of tight ion pairs in comparison to strong Lewis acids. However, it is not still clarified why the 1,4-enchainment is predominant with their catalysts.

4. DSC Analyses of Poly(CPD)s. The success in the controlled polymerization of CPD in terms of molecular weight and main-chain microstructure promoted him to investigate the thermal property of the polymers by differential scanning calorimetry (DSC) (Figure 7). The author employed three poly(CPD) samples with controlled molecular weights and narrow MWDs but low regiospecificity ($M_n = 2700, 6600, 10100; [1,4] = 55\%$ each) obtained with



Figure 8. Dependence of molecular weight and main-chain microstructure on T_g of poly(CPD). The three samples (\bullet : $M_n = 2700$, M_w/M_n = 1.17; $M_n = 6600$, $M_w/M_n = 1.26$; $M_n = 10100$, $M_w/M_n = 1.33$) with 45%-[1,2] and 55%-[1,4] were obtained with 1/SnCl₄/*n*-Bu₄NCl in CH₂Cl₂ at -78 °C: [CPD]₀ = 0.25, 0.50, or 1.00 M; [1]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu₄NCl]₀ = 5.0 mM. The sample (\blacktriangle : $M_n = 5100$, $M_w/M_n = 1.35$) with 30%-[1,2] and 70%-[1,4] were obtained with 1/ZnBr₂ in CH₂Cl₂ at -78 °C: [CPD]₀ = 0.50 M; [1]₀ = 5.0 mM; [ZnBr₂]₀ = 100 mM.

 $1/SnCl_4/n$ -Bu₄NCl, and another sample ($M_n = 5100$; [1,4] = 70%) with a higher regiospecificity with $1/ZnBr_2$ in CH₂Cl₂ at -78 °C.

Every sample showed an endothermic bias due to glass transition, and the glass transition temperature (T_g) was dependent on M_n and microstructure (Figure

7). The larger the M_n , the higher the T_g for the three samples with the same 1,4content. When the molecular weights are similar (6100 vs 5100), T_g decreased sharply with increasing 1,4-content, most likely due to a increase in the more flexible five-membered ring in the main chain.

In conclusion, enchainments in cationic polymerization of CPD with the 1/Lewis acid system can be affected not by solvents, additives, and temperatures but dramatically by Lewis acids. Specifically, weak Lewis acids such as $TiCl_2(Oi-Pr)_2$ and ZnX_2 (X = Cl, Br, I) induced regiospecific polymerization with relatively high 1,4-contents, among which the highest content was obtained with ZnI_2 ([1,4] = 76%).

EXPERIMENTAL

Materials

Unless otherwise specified, all of the chemicals were purchased from Aldrich. CPD was obtained by the retro Diels-Alder reaction of dicyclopentadiene (Tokyo Kasei; >95%) over calcium hydride and distilled from calcium hydride.⁷ It was stored at -84 °C and used within one month. The purity and the absence of dicyclopentadiene in CPD was confirmed by NMR before use (purity >99%). CH₂Cl₂ (Wako; >99%) was dried overnight over calcium chloride, distilled from pentoxide and phosphorous then from calcium hydride before use. Methylcyclohexane (Tokyo Kasei; >99%), carbon tetrachloride (Wako; internal standard for gas chromatography; >99%), ethyl acetate (Wako; >99%) and 1butylimidazole (98%) were dried overnight over calcium chloride, distilled twice Toluene (Wako; >99%) and diethyl ether (Wako; >99%) from calcium hydride. were dried overnight over calcium chloride, distilled from sodium benzophenone

ketyl. The following materials were used as received: SnCl₄, SnBr₄ and BCl₃ (all 1.0 M solution in CH₂Cl₂); TiCl₄ (>99.9%); TiBr₄ and AlBr₃ (both >99.99%); ZnCl₂ (1.0 M solution in Et₂O); ZnBr₂ and ZnI₂ (both >99.999%); CH₃SO₃H (Nacalai Tesque; 99%); *n*-Bu₄NCl and Ph₃CCl (both Tokyo Kasei; >98%); 2,6-di*tert*-butyl-4-methylpyridine (98%); 2,4,6-trimethylpyridine (99%); and isopropyl ether (anhydrous; 99%). TiCl₂(O*i*-Pr)₂ was prepared and purified as already reported.²⁵ The HCl adduct 1 was synthesized by bubbling dry HCl gas into a solution of CPD as already reported.^{7,26}

Polymerization Procedures

Polymerization was carried out under dry nitrogen in baked glass tubes equipped with a three-way stopcock. A typical example for the polymerization procedure with the 1/ZnBr₂ system is given below: The reaction was initiated by sequential addition of prechilled solutions of 1 (0.015 mmol; 0.30 mL of 0.050 M in CH₂Cl₂) and ZnBr₂ (0.3 mmol; 0.30 mL of 1.0 M in Et₂O) via dry syringes into a monomer solution (in CH₂Cl₂; 2.4 mL) containing CPD (1.5 mmol; 0.124 mL) and CCl₄ (0.124 mL). The total volume of the reaction mixture was thus 3.0 mL. For polymerization in the presence of an additive, it was dissolved in the Lewis acid solution prior to initiation. After predetermined intervals, the polymerization was terminated with prechilled methanol (2.0 mL) containing a small amount of ammonia. 3-tert-Butyl-4-hydroxy-5-methylphenyl sulfide (0.014 mmol; 0.005 g) was then added as an antioxidant immediately after quenching reaction. Monomer conversion was determined from its residual concentration measured by gas chromatography with CCl₄ as an internal standard. The polymer yield by gravimetry was in good agreement with the gas-chromatographic conversion of the monomer. The quenched reaction mixture was washed with dilute hydrochloric acid and with water to remove initiator residues, evaporated to dryness under reduced pressure, and vacuum dried to give the product polymers.

Measurements

The MWD of the polymers were measured by size-exclusion chromatography (SEC) in chloroform at 40 °C on three polystyrene gel columns [Shodex K-805L (pore size: 20–1000 Å; 8.0 nm i.d. × 30 cm) × 3; flow rate 1.0 mL/min] that were connected to a Jasco PU–980 precision pump and a Jasco 930-RI refractive index detector. The columns were calibrated against 11 standard polystyrene samples (Pressure Chemicals; $M_n = 580-1547000$; $M_w/M_n \le 1.1$) as well as the styrene monomer. ¹H NMR spectra of the produced polymers were recorded in CDCl₃ on a JEOL LNM–LA500 spectrometer, operating at 500.2 MHz. The DSC measurements were performed on a Thermo Plus 2 Series (Rigaku, Tokyo) equipped with a DSC8230L module. Polymer solid samples (ca. 6 mg) were measured in an aluminum container under a dry nitrogen flow at a heating or cooling rate of 10 deg/min. α -Alumina was used as a standard.

REFERENCES AND NOTES

- For a review on cationic polymerization; see: Kennedy, J. P. Cationic Polymerization of Olefins: A Critical Inventory, Wiley-Interscuence, New York, 1975.
- (2) Staudinger, H.; Bruson, H. A. Ann. 1926, 447, 110.
- (3) Aso, C.; Kunitake, T.; Ishimoto, Y. J. Polym. Sci. Part A-1 1968, 6, 1163.
- (4) Sigwalt, P.; Vairon, J. P. Bull. Soc., Chim. France 1964, 482.
- (5) Momiyama, Z.; Imanishi, Y.; Higashimura, T. Chem. High Polymers, 1966, 23, 56.
- (6) Aso, C.; Kunitake, T.; Ishimoto, Y. J. Polym. Sci. Part A-1 1968, 6, 1175.
- (7) Chapter 4 of this thesis; Ouchi, M.; Kamigaito, M.; Sawamoto, M. Macromolecules, in press.

- (8) McCann, M.; Coda, E. M. G. J. Mol. Catal.: A. Chem. 1996, 109, 99.
- (9) Kühn, F. E.; Ismeier, J. R.; Schon, D.; Xue, W.-M.; Zhang, G.; Nuyken, O. Macromol. Rapid Commun. 1999, 20, 555.
- (10) Kaminsky, W.; Bark, A.; Spiehl, R.; Moller-Lindenhof, N.; Nieboda, S. In *Transition Metals and Organometallics as Catalysts for Olefin Polymerization*; Kaminsky, W., Sinn, H., Eds.; Springer-Verlag: Berlin, 1988; p 291.
- (11) Arndt, M.; Kaminsky, W. Macromol. Symp. 1995, 95, 167.
- (12) Arndt, M.; Kaminsky, W. Macromol. Symp. 1995, 97, 225.
- (13) Collins, S.; Kelly, W. M. Macromolecules 1992, 25, 233.
- (14) Kelly, W. M.; Taylor, N. J.; Collins, S. Macromolecules 1994, 27, 4477.
- (15) Kelly, W. M.; Wang, S. T.; Collins, S. Macromolecules 1997, 30, 3151.
- (16) McLain, S. J.; Feldman, J.; McCord, E. F.; Gardner, K. H.; Teasley, M. F.; Coughlin, E. B.; Sweetman, B. J.; Johnson, L. K.; Brookhart, M. *Macromolecules* 1998, 31, 6705.
- (17) For a reveie on living anionic polymeirzations of dienes; see: Szwarc, M. Carbanions, Living Polymers, and Electron-Transfer Process, John Wiley & Sons, New York, 1968, p 513.
- (18) For a review on cationic polymerizations of dienes; see: Nuyken, O.; Pask, Std. D. In *Comprehensive Polymer Science*, Eastmond, G. C., Ledwith, A., Russo, S., Sigwalt, P., Eds.; Vol. 3; Pergamon: Oxford, 1989; p 619.
- (19) Chapter 1 of this thesis; Ouchi, M.; Kamigaito, M.; Sawamoto, M. Macromolecules 1999, 32, 6407.
- (20) Higashimura, T.; Yonezawa, T.; Okamura, S.; Fukui, K. J. Polym. Sci. 1959, 39, 487.
- (21) Kunitake, T.; Aso, C. J. Polym. Sci., Part A-1 1970, 8, 665.
- (22) Kishimoto, Y.; Aoshima, S.; Higashimura, T. Macromolecules 1989, 22, 3877.

- (23) Kamigaito, M.; Maeda, Y.; Sawamoto, M.; Higashimura, T. Macromolecules 1993, 26, 1643.
- (24) Polymerization conditions (the same as those in Table 2, footnote *a*): [CPD]₀ = 500 mM; [CH₃SO₃H]₀ = 5.0 mM; at -78 °C in CH₂Cl₂; time 20 min; conversion = 78 %; M_n = 37,000; M_w/M_n = 4.20 (bimodal); [1,4] = 51%; D/(A+B+C) = 0.49.
- (25) Kamigaito, M.; Sawamoto, M.; Higashimura, T. Macromolecules 1995, 28, 5671.
- (26) Katayama, H.; Kamigaito, M.; Sawamoto, M. Macromolecules 1995, 28, 3747.
Synthesis of End-Functionalized CPD Polymers and Copolymers of CPD

ABSTRACT

A series of cyclopentadiene (CPD)-based polymers and copolymers were synthesized by a controlled cationic polymerization of CPD. End-functionalized synthesized with the HCl-adducts [initiator: poly(CPD) was $CH_3CH(OCH_2CH_2X)Cl; X = Cl (2a)$, acetate (2b), methacrylate (2c)] of vinyl ethers carrying pendant functional substituents X in conjunction with SnCl₄ (Lewis acid as catalyst) and n-Bu₄NCl (additive) in dichloromethane at -78 °C. The system led to controlled cationic polymerizations of CPD to give controlled α -endfunctionalized poly(CPD)s with almost quantitative attachment of the functional groups ($F_n \sim 1$). With the 2a or 2b/SnCl₄/n-Bu₄NCl initiating systems, diblock copolymers of 2-chloroethyl vinyl ether (CEVE) and 2-acetoxyethyl vinyl ether (AcOVE) with CPD were also synthesized by sequential polymerization of CPD and these vinyl ethers. ABA-Type triblock copolymer of CPD (A) and CEVE (B) was also prepared with the use of a bifunctional initiator. Copolymerization of CPD and CEVE with 2a/SnCl4/n-Bu4NCl afforded random copolymers with controlled molecular weights and narrow molecular weight distributions $(M_w/M_n =$ 1.3–1.4).

INTRODUCTION

As discussed in Chapter 4, the author has succeeded in controlling polymer molecular weight and its distribution in cationic polymerization of cyclopentadiene (CPD),¹ even though conventional cationic systems for CPD usually result in oligomers of ill-defined molecular weights and structures. CPD is an inexpensive and attractive monomer that gives rigid polymers of cyclic hydrocarbon repeat units with attractive thermal, mechanical, and electronic properties such as high glass-transition temperature and low dielectric constant. The new CPD polymerization is achieved with an initiating system consisting of an initiator, an activator (catalyst), and an additive (eq 1). The initiator may be the HCl–adduct of CPD (1) or 2-chloroethyl vinyl ether (CEVE) (2a) (initiator); the catalyst is a Lewis acid such as SnCl4; typical additives include *n*-Bu₄NCl, diethyl ether, and ethyl acetate. As in living polymerization, one initiator generates one polymer chain where the initiator moiety is attached to the α -end (eq 1).



This chapter is directed to demonstrate the synthetic capability of the new CPD polymerization. The first purpose was thus the synthesis of end-functionalized poly(CPD)s (3) with the use of the HCl-adducts (2b and 2c) of functional vinyl ethers as initiators (eq 2). These adducts are effective functional initiators in living cationic polymerization of vinyl ethers², styrene derivatives³ and β -pinene⁴ to give α -end-functionalized polymers. Specifically, 2b can be converted into the corresponding hydroxy group by deprotection of the acetoxy group. Macromonomers with a methacryloxy group can be obtained from 2c.

-108-

End-Functionalized Polymers and Copolymers



The second objective was to synthesize block and random copolymers of CPD with vinyl ethers. For this pair of monomers, the CPD unit can be regarded as a hard segment, whereas the vinyl ether segment as a soft segment with polar ether substituents. Structural formulae 4–7 illustrate the target polymers. AB-block polymers (4 and 5) involve 2-chloroethyl vinyl ether (CEVE) and 2-acetoxyethyl vinyl ether (AcOVE). Upon hydrolysis,⁵ the acetoxyethyl moiety in 5 may lead to new amphiphilic block copolymers of CPD. The ABA-type triblock copolymer (6), consisting of hard and soft segments, is also obtained from a bifunctional initiator. In addition, the controlled random copolymerization of CPD and CEVE was investigated (cf. Formula 7).



Formulae 4-7

-109-

Chapter 6

RESULTS AND DISCUSSIONS

1. End-Functionalized Poly(CPD)s

As shown in eq 2, end-functionalized poly(CPD)s (**3b** and **3c**) were prepared with **2b** and **2c**, respectively, in CH₂Cl₂ at -78 °C. Initiators **2b** and **2c** induced slow and incomplete polymerizations under the conditions ([SnCl₄]₀ = [*n*-Bu₄NCl]₀ = 5.0 mM) suitable for the controlled polymerization with **2a**.¹ This is probably due to the ester groups in **2b** and **2c**, which may interact with SnCl₄ and weaken its Lewis acidity. However, a lower concentration of *n*-Bu₄NCl (4.0 mM) permitted quantitative polymerizations, though slower than that with **2a** (Figure 1).



Figure 1. Time-conversion curves for the polymerization of CPD with $R-Cl/SnCl_4/n$ -Bu_4NCl in CH₂Cl₂ at -78 °C: [M]₀ = 0.50 M; [R-Cl]₀ = 5.0 mM; [SnCl₄]₀ = 5.0 mM; [*n*-Bu_4NCl]₀ = 4.0 mM. $R-Cl: (\blacklozenge) 2a; (\spadesuit) 2b; (\blacktriangle) 2c.$

Figure 2 shows the number-average molecular weight (M_n) and molecular weight distributions (MWDs) of the obtained polymers obtained from 2b and 2c, as

End-Functionalized Polymers and Copolymers



Figure 2. M_n , M_w/M_n , and MWD curves of poly(CPD) obtained with R-Cl/SnCl₄/*n*-Bu₄NCl in CH₂Cl₂ at -78 °C: $[M]_0 = 0.50$ M; $[R-Cl]_0 = 5.0$ mM; $[SnCl_4]_0 = 5.0$ mM; $[n-Bu_4NCl]_0 = 4.0$ mM. R-Cl: (\bullet , O) 2b; (\blacktriangle , \triangle) 2c.

measured by size-exclusion chromatography (SEC) with a polystyrene calibration. The M_n increased in direct proportion to monomer conversion, and the MWDs were narrow throughout the reactions ($M_w/M_n = 1.2-1.5$). Thus, both the acetoxy and methacryloxy-functionalized initiators led to CPD polymers with controlled molecular weights and narrow MWDs.

The end-groups of these polymers were analyzed by ¹H NMR spectroscopy. Figure 3A shows the spectrum of the poly(CPD) (**3b**) obtained from initiator **2b**. The sample shows large characteristic signals of the main chain olefinic (*d* and *h*) and aliphatic protons (*a*-*c* and *e*-*g*). Additionally, small absorptions appeared, assignable to the initiator fragment (*i*-*m*). The M_n (NMR) was thus determined from the relative peak area of the main-chain repeat units (*d* and *h*) to the initiator moiety (*l*). The value for the sample was close to that by SEC [M_n (NMR) = 2100, M_n (SEC) = 2600], indicating that **2b** forms one polymer chain per molecule.

Figure 3B shows the ¹H NMR spectrum of the methacryloxy-end poly(CPD) (3c) from 2c. The spectrum is very similar to Figure 3A except for the

-111-



Figure 3. ¹H NMR spectra (500 MHz, CDCl₃) of poly(CPD)s [(A) $M_n = 2600$, $M_w/M_n = 1.18$, (B) $M_n = 2500$, $M_w/M_n = 1.20$] obtained with **2b** (A) and **2c** (B). See Figure 2 for reaction conditions.

absorptions of the methyl (*n*) and the vinyl proton (*o*) of the methacryloxy group. The M_n (NMR) was also close to that by SEC [M_n (NMR) = 2400, M_n (SEC) = 2500]. Thus, the functionalized initiators derived from the corresponding vinyl ethers proved effective in the synthesis of end-functionalized poly(CPD)s.

2. Block Copolymerization of CPD with Vinyl Ethers

2-1. Diblock Copolymers

The AB-block copolymer 4 was synthesized by sequential polymerization of VE and CPD with the HCl-vinyl ether adduct/SnCl₄/*n*-Bu₄NCl system. The first step was the living cationic polymerization of CEVE with 2a in CH₂Cl₂ at -78 °C. CPD equimolar to the CEVE was then added. In this system, the living CEVE polymer can be regarded as a macroinitiator with a C-Cl terminal, similar to 2a.



Figure 4. MWD curves of (A) poly(CEVE), (B) CEVE-CPD block copolymer 4, (C) poly(AcOVE), and (D) AcOVE-CPD block copolymer 5. Poly(CEVE) and 4 were obtained with $2a/SnCl_4/n$ -Bu_4NCl in CH₂Cl₂ at -78 °C: [CEVE]₀ = [CPD]_{add} = 0.50 M; [2a]₀ = 5.0 mM; [SnCl₄]₀ = 10 mM; [*n*-Bu_4NCl]₀ = 9.0 mM. Poly(AcOVE) and 5 were obtained with $2b/SnCl_4$ in CH₂Cl₂ at -78 °C: [AcOVE]₀ = [CPD]_{add} = 0.50 M; [2b]₀ = 5.0 mM; [SnCl₄]₀ = 20 mM.

The polymerization of CEVE proceeded in a living fashion to form a poly(CEVE) prepolymer with a very narrow MWD ($M_w/M_n = 1.07$) (Figure 4A). The subsequent polymerization of CPD proceeded smoothly and the SEC curve of the product shifted to higher molecular weight relative to the CEVE prepolymer and



Figure 5. ¹H NMR spectra (500 MHz, CDCl₃) of (A) poly(CEVE) ($M_n = 10600, M_w/M_n = 1.08$), (B) CEVE-CPD block copolymer ($M_n = 19500, M_w/M_n = 1.20$), (C) poly(AcOVE) ($M_n = 12800, M_w/M_n = 1.06$), and (D) AcOVE-CPD block copolymer ($M_n = 20100, M_w/M_n = 1.23$).

the MWD curves stayed narrow $(M_w/M_n = 1.20)$ (Figure 4B). These indicate the formation of an AB-block copolymer of CEVE and CPD.

Block copolymer (5) of AcOVE and CPD was also synthesized in a similar way. AcOVE was first polymerized with initiator 2b. The polymerization was living even in the absence of an additive (Figure 4C), because of the ester moiety in the monomer.⁶ This in tern requires a higher concentration of SnCl₄ (20 mM) for quantitative polymerizations. In the second step, CPD polymerized in a controlled fashion from the living AcOVE and under the additive free condition, to give a block copolymer with a narrow MWD ($M_w/M_n = 1.17$). Thus, the ester group in AcOVE unit serves as an effective basic additive for controlling the polymerization of CPD as well.

Figure 5A shows the ¹H NMR spectrum of the prepolymer of CEVE after quenching with methanol, where small absorptions originating from the α - (a) and ω -end groups (f and g) are observed in addition to signals of the main chain repeat units (b-e). The spectrum of the final product (Figure 5B) exhibits the typical signals of CPD repeat units, olefinic (k and o) and aliphatic protons (h-j and l-n), along with the absorptions of poly(CEVE) prepolymer. The terminal methine proton (f) of CEVE prepolymer is absent in the spectrum of 4 (Figure 5B). These indicate that every CEVE polymer chain initiated the intended block copolymer. The ratio of CEVE and CPD units [n/(m+l)] in 4 was 100/84 [by NMR; (c+d+e)/(k+o)], in good agreement with the calculated value 100/89, based on the initial feed ratio and conversion.

A similar NMR analysis also confirmed the formation of the AcOVE-CPD block polymers (Figure 5C and 5D). The AcOVE/CPD units ratio n/(m+1) [= e/(l+p) by NMR] was 97/85 close to the calculated 97/94.

2-2. Synthesis of Triblock Copolymer

The clean formation of AB-type diblock copolymers of VE and CPD was then extended to ABA-type triblock copolymers. In particular, CPD-VE-CPD



Scheme 1. Synthesis of CPD-CEVE-CPD Triblock Copolymer with a Bifunctional Initiator



Figure 6.

MWD curves of (A) poly(CEVE) and (B) CPD-CEVE-CPD triblock copolymer (6) obtained with $8/SnCl_4/n$ -Bu₄NCl in CH₂Cl₂ at -78 °C: [CEVE]₀ = 0.50 M; [CPD]_{add} = 1.0 M; [8]₀ = 5.0 mM; [SnCl₄]₀ = 20 mM; [*n*-Bu₄NCl]₀ = 18 mM. triblock copolymer can be a thermoplastic elastomer. The author examined the synthesis by sequential polymerization from VE to CPD via a bifunctional initiator (8) (Scheme 1).

The polymerization of CEVE with 8 gave a bifunctional living prepolymer of CEVE with a narrow MWD ($M_w/M_n = 1.20$) (Figure 6A). After the consumption of CEVE, a double molar amount of CPD to that of CEVE was added to the reaction mixture. CPD was smoothly polymerized (93% conversion in 5 h), and the SEC curve of the product shifted to higher molecular weight ($M_w/M_n = 1.34$), and the prepolymer seems absent (Figure 6B).

Figure 7A shows the ¹H NMR spectrum of the two-armed prepolymer of CEVE before the CPD addition. Key signals include these of the aromatic protons (*b* and *c*) from **8** and the terminal methine (*l*) and methoxy (*m*) protons. The arm number (F_n) of the polymer was 2.0, calculated from the peak intensity





ratio of the terminal methine (l) to the aromatic protons (b and c). This result shows that the initiator 8 gave two-armed living poly(CEVE).

The final product (Figure 7B) after the CPD addition exhibit the signals of olefinic (q and u) and aliphatic protons (n-p and r-t), characteristic to CPD, and the terminal methine (l) disappeared. The CEVE/CPD units ratio [n/(m+l) = (i+j+k)/(q+u)] was 100/180 (calcd.,100/186). These results demonstrated the formation of the CPD-CEVE-CPD triblock copolymer (6). The product was sticky like an uncrosslinked rubber and different from the respective homopolymers and the CEVE-CPD diblock copolymer (4).

3. Random Copolymerization of CPD and CEVE

An equimolar mixture of CPD and CEVE was polymerized with $2a/SnCl_4/n$ -Bu₄NCl in CH₂Cl₂ at -78 °C (Scheme 2). As seen in Figure 8, both monomers were consumed almost simultaneously, more slowly than in the respective homopolymerizations.



CPD-CEVE Random Copolymer (7)

Scheme 2

The M_n of the products increased with the yield. The MWDs were unimodal and relatively narrow throughout the reactions $(M_w/M_n = 1.3-1.4)$. Thus, the controlled random copolymerization of CPD and CEVE proved possible with 2a/SnCl₄/*n*-Bu₄NCl, where copolymer composition is nearly close to the initial mole ratio of the monomers, even at low conversion.

The copolymer was apparently random as examined by ¹H NMR (Figure 9). The spectrum of the product (Figure 9C) was different from those of the



Figure 8. Copolymerization of CPD (\blacktriangle) and CEVE (\bigcirc) and homopolymerization of CPD (\blacksquare) and CEVE (\diamondsuit) with 2a/SnCl₄/n-Bu₄NCl in CH₂Cl₂ at -78 °C: [CPD]₀ = [CEVE]₀ =0.50 M; [2a]₀ = 5.0 mM; [SnCl₄]₀ = 10 mM; [n-Bu₄NCl]₀ = 9.0 mM.

homopolymers (Figure 9A and 9B), their blend (Figure 9D), and the CEVE-CPD block copolymer (Figure 5B). The peaks of the copolymer (7) were broader than those of the others for both CPD and CEVE units, which supports the random sequence of the two units. The ratio of CPD and CEVE units [(n+m)/l] was 44/36 (calculated 44/42).



Figure 9. ¹H NMR spectra (500 MHz, CDCl₃) of (A) poly(CPD) ($M_n = 7000$, $M_w/M_n = 1.44$), (B) poly(CEVE) ($M_n = 8600$, $M_w/M_n = 1.07$), (C) CPD-CEVE random copolymer ($M_n = 12600$, $M_w/M_n = 1.41$), and (D) blend of poly(CPD) and poly(CEVE).

EXPERIMENTAL

Materials

Unless otherwise specified, all of the chemicals were purchased from Wako company, Japan. CPD was obtained by the retro Diels-Alder reaction of

dicyclopentadiene (Tokyo Kasei; >95%) over calcium hydride and distilled from calcium hydride. It was stored at -84 °C and used within one month. The purity and the absence of dicyclopentadiene in CPD was confirmed by NMR before use (purity >99%). 2-chloroethyl vinyl ether (CEVE) (Nisso Maruzen Chemical; > 99%) was washed with 10% aqueous sodium hydroxide and then water, dried overnight with anhydrous sodium sulfate, and distilled twice under reduced pressure from calcium hydride before use. 2-Acetoxy vinyl ether $(AcOVE)^7$ and 2-vinyloxyethyl methacrylate (VEM)⁸ (precursor for 2c) were prepared by the reaction of CEVE with sodium acetate and sodium methacrylate respectively, dried overnight with anhydrous sodium sulfate, and distilled twice under reduced pressure from calcium hydride before use. CH₂Cl₂ (>99%) was dried overnight over calcium chloride, distilled from phosphorous pentoxide and then from calcium hydride before use. Carbon tetrachloride (internal standard for gas chromatography; >99%) was dried overnight over calcium chloride, distilled twice from calcium hydride. SnCl₄ (Aldrich; 1.0 M solution in CH₂Cl₂), n-Bu₄NCl (Tokyo Kasei; >98%), and bisphenol A divinyloxyethyl ether (BPA-DEVE) (Nisso Maruzen Chemical; > 99%; precursor for 3) were used as received.

Synthesis of HCl Adduct 2a-2c and 8

The HCl adducts 2a-2c and 8 were synthesized by bubbling dry HCl gas into a solution of the respective vinyl compounds (CEVE, AcOVE, VEM and BPA-DEVE).⁹ The synthesis of HCl-CPD adduct 2a was carried out as follows. Dry HCl gas was bubbled into 1.0 M CH₂Cl₂ solution (35 mL) of CEVE (4.0 mL; 0.039 mol) at -78 °C. The gas was generated by dropping concentrated sulfuric acid into powdery sodium chloride, and dried by passing it through a column packed with calcium chloride. After 30 min, the excess HCl in the reaction mixture was removed by bubbling dry nitrogen gas at 0 °C for 30 min, and the clean and quantitative formation of adduct 2a was confirmed by ¹H NMR spectroscopy. The concentration of 2a was determined by titrating the chloride by the Volhard method. The observed value (1.03 M) further confirmed the quantitative formation of 2a from CEVE.

Polymerization Procedures

Polymerization was carried out under dry nitrogen in baked glass tubes equipped with a three-way stopcock. A typical example for the polymerization procedure for α -end-functionalyzed polymer is given below: The reaction was initiated by sequential addition of prechilled solutions of 2b (0.015 mmol; 0.30 mL of 0.050 M in CH₂Cl₂) and a mixture of SnCl₄ (0.015 mmol; 0.30 mL of 0.050 M in CH₂Cl₂) and *n*-Bu₄NCl (0.012 mmol; 0.30 mL of 0.040 M in CH₂Cl₂) via dry syringes into a monomer solution (in CH₂Cl₂; 2.4 mL) containing CPD (1.5 mmol; 0.124 mL) and CCl₄ (0.124 mL). The total volume of the reaction mixture was thus 3.0 mL. After predetermined intervals, the polymerization was terminated with prechilled methanol (2.0 mL) containing a small amount of ammonia. Another typical example for the block copolymerization of CEVE-CPD is given below: The reaction was initiated by sequential addition of prechilled solutions of 2a (0.015 mmol; 0.30 mL of 0.050 M in CH₂Cl₂) and a mixture of SnCl₄ (0.015 mmol; 0.30 mL of 0.050 M in CH₂Cl₂) and n-Bu₄NCl (0.015 mmol; 0.30 mL of 0.050 M in CH₂Cl₂) via dry syringes into a monomer solution (in CH₂Cl₂; 2.4 mL) containing CEVE (1.5 mmol; 0.153 mL) and CCl₄ (0.153 mL). The total volume of the reaction mixture was thus 3.0 mL. The reaction was run for 1 min, where CEVE conversion reached 100%. CPD (1.5 mmol; 0.124 mL) was then added to the preformed living poly(CEVE) solution. 3-tert-Butyl-4-hydroxy-5methylphenyl sulfide (0.014 mmol; 0.005 g) was then added as an antioxidant immediately after quenching reaction containing poly(CPD) or copolymer with CPD units. Monomer conversion was determined from its residual concentration measured by gas chromatography with CCl₄ as an internal standard. The polymer yield by gravimetry was in good agreement with the gas-chromatographic

conversion of the monomer. The quenched reaction mixture was washed with dilute hydrochloric acid and with water to remove initiator residues, evaporated to dryness under reduced pressure, and vacuum dried to give the product polymers.

Measurements

The MWD polymers size-exclusion of the were measured by chromatography (SEC) in chloroform at 40 °C on three polystyrene gel columns [Shodex K-805L (pore size: 20–1000 Å; 8.0 nm i.d. \times 30 cm) \times 3; flow rate 1.0 mL/min] that were connected to a Jasco PU-980 precision pump and a Jasco 930-RI refractive index detector. The columns were calibrated against 11 standard polystyrene samples (Pressure Chemicals; $M_{\rm n} = 580-1547000; M_{\rm w}/M_{\rm n} \le 1.1$) as well as the styrene monomer. ¹H NMR spectra of the produced polymers were recorded in CDCl₃ on a JEOL LNM-LA500 spectrometer, operating at 500.2 MHz.

REFERENCES AND NOTES

- Chapter 4 of this thesis; Ouchi, M.; Kamigaito, M.; Sawamoto, M. Macromolecules, in press.
- (2) (a) Aoshima, S.; Ebara, K.; Higashimura, T. Polym. Bull. 1985, 14, 425. (b) Sawamoto, M.; Enoki, T.; Higashimura, T. Macromolecules 1987, 20, 1.
- (3) (a) Shohi, H.; Sawamoto, M.; Higashimura, T. *Maclomolecules* 1992, 25, 53.
 (b) Shohi, H.; Sawamoto, M.; Higashimura, T. *Maclomol. Chem.* 1992, 193, 1783. (c) Miyashita, K.; Kamigaito, M.; Sawamoto, M.; Higashimura, T. *Maclomolecules* 1994, 27, 1093. (d) Sawamoto, M.; Hasebe, T.; kamigaito, M.; Higashimura, T. J. *Macromol. Sci.-Pure Appl. Chem.* 1994, A31, 937.

- (4) Lu, J.; Kamigaito, M.; Sawamoto, M.; Higashimura, T.; Deng, Y.-X. J. Polym. Sci., Part A: Polym. Chem. 1997, 35, 1423.
- (5) Minoda, M.; Sawamoto, M.; Higashimura, M. Macromolecules 1987, 20, 2045.
- (6) Aoshima, S.; Higashimura, T. Polym. Bull. 1986, 15, 417. Aoshima, S.; Nakamura, T.; Uesugi, N.; Sawamoto, M.; Higashimura, T. Macromolecules 1985, 18, 207.
- (7) Aoshima, S.; Nakamura, T.; Uesugi, N.; Sawamoto, M.; Higashimura, T. Macromolecules 1985, 18, 207.
- (8) Aoshima, S.; Hasegawa, O.; Higashimura, T. Polym. Bull. 1985, 13, 229.
- (9) Katayama, H.; Kamigaito, M.; Sawamoto, M. Macromolecules 1995, 28, 3747.

LIST OF PUBLICATIONS

- Chapter 1 Macromolecules 1999, 32, 6407.
- Chapter 2 J. Polym. Sci., Part A: Polym. Chem. 2001, 39, 1060.
- Chapter 3 J. Polym. Sci., Part A: Polym. Chem. 2001, 39, 1067.

Chapter 4 *Macromolecules*, in press.

Chapter 5 *Macromolecules*, submitted.

Chapter 6 J. Polym. Sci., Part A: Polym. Chem. 2001, 39, 398.

Other Publications

"Amphiphilic 3-Arm Star Block polymers by Living Cationic Polymerization" Takahashi, R.; Ouchi, M.; Satoh K.; Kamigaito, M.; Sawamoto, M. Polym. J. **1999** 31 995.

"MALDI-TOF-MS Analysis of Ru(II)-Mediated Living Radical Polymerizations of Methyl Methacrylate, Methyl Acrylate, and Styrene" Nonaka, H.; Ouchi, M.; Kamigaito, M.; Sawamoto, M. *Macromolecules*, in press.

ACKNOWLEDGMENTS

This thesis presents the studies which the author carried out from 1995 to 2000 at the Department of Polymer Chemistry of Kyoto University under the direction of Professor Mitsuo Sawamoto.

The author would like to express his deep gratitude to Professor Mitsuo Sawamoto for his continuous guidance and encouragement throughout the course of this work. A grateful acknowledgment is due to Associate Professor Masami Kamigaito for his helpful suggestions and stimulating discussions. Very sincere thanks are due to Mr. Hiroshi Katayama and Dr. Kotaro Satoh for their kind guidance in experimental techniques and discussions in the early phase of this research. He also wishes to express his thanks to all the members of his group in the five and half years for their kind help and encouragement.

And the author gratefully acknowledges support for this work by a grant from the Japan Society for the Promotion Sciences (JSPS) for JSPS Research Fellowships for Young Scientists.

Finally, the author heartily thanks his family for their care and encouragement.

January 2001

M. Oucho

MAKOTO OUCHI

Department of Polymer Chemistry Kyoto University Kyoto, Japan